## Texas Southern University Digital Scholarship @ Texas Southern University

**Dissertations (2016-Present)** 

**Dissertations** 

8-2022

## An Examination of Public-School Discipline Policies and Implications for African American Students within a Texas School District

Demeka Demeka Simmons

Follow this and additional works at: https://digitalscholarship.tsu.edu/dissertations Part of the Public Administration Commons, and the Public Affairs Commons

#### **Recommended Citation**

Simmons, Demeka Demeka, "An Examination of Public-School Discipline Policies and Implications for African American Students within a Texas School District" (2022). *Dissertations (2016-Present)*. 1. https://digitalscholarship.tsu.edu/dissertations/1

This Dissertation is brought to you for free and open access by the Dissertations at Digital Scholarship @ Texas Southern University. It has been accepted for inclusion in Dissertations (2016-Present) by an authorized administrator of Digital Scholarship @ Texas Southern University. For more information, please contact haiying.li@tsu.edu.

# AN EXAMINATION OF PUBLIC-SCHOOL DISCIPLINE POLICIES AND IMPLICATIONS FOR AFRICAN AMERICAN STUDENTS WITHIN A TEXAS SCHOOL DISTRICT

DISSERTATION

Presented in Partial Fulfillment of the Requirements for

the Degree Doctor of Philosophy in the Graduate School

of Texas Southern University

By

Demeka Simmons, B.S., M.S.

Texas Southern University

2022

Approved By

Dr. David Baker Chairperson, Dissertation Committee

Dr. Jasmine Drake Co-Chairperson, Dissertation Committee

Dr. Gregory H. Maddox Dean, The Graduate School Approved By

Dr. David Baker	<u>06/27/22</u>
Chairperson, Dissertation Committee	Date
Dr. Jasmine Drake	<u>06/27/22</u>
Co-Chairperson, Dissertation Committee	Date
Dr. Howard Henderson	<u>06/29/22</u>
Committee Member	Date
Dr. Danita Bailey-Samples	<u>06/28/22</u>
Committee Member	Date

© Copyright Demeka Simmons 2022 All Rights Reserved

# AN EXAMINATION OF PUBLIC-SCHOOL DISCIPLINE POLICIES AND IMPLICATIONS FOR AFRICAN AMERICAN STUDENTS WITHIN A TEXAS SCHOOL DISTRICT

Demeka Simmons, Ph.D.

Texas Southern University, 2022

Professor David Baker, Advisor

National data illustrates that the use of exclusionary discipline practices has resulted in disproportionality and overrepresentation in school discipline. Additionally, research also illustrates that African Americans have been overrepresented in exclusionary discipline at higher rates than their White counterparts. The purpose of this study was to determine if there was a statistically significant difference in the percent of exclusionary discipline practices between African American students and White students based on school district setting within Texas. The researcher analyzed archival data from 673 students in grades 3-12 enrolled in a Texas public school. The statistical tests used in the analysis of the research questions included Chi-Square and multinomial logistic regression analysis. The findings indicated that there was no significant relationship between student race and exclusionary disciplinary practices. Based on the results, a significant relationship was found to exist between student gender and exclusionary discipline practices. After controlling for gender, the findings revealed that there was no significant relationship between race and occurrences of exclusionary discipline practices of the school district in the study. However, in hypothesis three-part b controlling for race, there was a significant relationship between gender and occurrences of exclusionary discipline practices.

Keywords: *exclusionary discipline, disciplinary alternative education program, zero tolerance* 

### TABLE OF CONTENTS

LIST C	F TABLES	iv
VITA		v
DEDIC	ATION	vi
ACKN	OWLEDGEMENTS	vii
CHAP	TER	
1.	INTRODUCTION	1
2.	LITERATURE REVIEW	17
3.	DESIGN OF THE STUDY	54
4.	RESULTS	58
5.	DISCUSSION	71
REFER	ENCES	88

### LIST OF TABLES

Table		Page
1.	Frequency Characteristics by Gender and Race	59
2.	Discipline Type	60
3.	Number and Percent of Exclusionary Practice by Race	60
4.	Crosstabulation for Race versus Exclusionary Discipline Practices	61
5.	Chi-square test for Race versus Exclusionary Discipline Practices	61
6.	Number and Percent of Exclusionary Practice by Gender	62
7.	Cross-tabulation for Gender versus Exclusionary Discipline Practices	63
8.	Chi-square test for Gender versus Discipline	63
9.	Model Fitting Information and Pseudo R-Square	64
10.	Likelihood Ratio Tests	66
11	Parameter Estimates for the Multinomial Logistic Regression Model	67
12.	Summary of All Hypotheses Tested	69

## VITA

2008	Bachelor of Administration of Criminal Justice Mountain State University Beckley, WV
2010	Master of Criminal Justice Administration Mountain State University Beckley, WV
2006 - 2015	CTE Teacher 9th-12th Grade MacArthur High School Houston, TX
2011 - 2015	Criminal Justice Adjunct Instructor Ashford University Tampa, Florida
2015 - 2017	Assistant Principal 9th -12th Grade Huntsville High School Huntsville, TX
2017 - 2019	Assistant Principal-Interim Principal 9th -12th Grade Furr High School Houston, TX
2019 - 2020	Principal 6th-8th Grade Shepherd Middle School Shepherd, TX
2020 - 2021	Principal 6th-8th Grade Marlin Middle School Marlin, TX
2021 - 2022	Principal 9th-12th Manor Excel Academy Manor, TX
Major Field	Administration of Justice

#### **DEDICATION**

I dedicate this dissertation to my strength, my bliss, my Lord and Savior, Jesus Christ. His unyielding grace and mercy over my life birthed tenacity and resilience to persevere in this journey through all the obstacles that have occurred during this journey we call life. I continue to give you all the honor and glory.

To the Simmons family, especially my virtuous grandmother Georgia Bell (McGowen) Simmons, who sacrificed her dreams, education, and so much more to commit her life to her children, grandchildren, and generations to come, this is for you. To my children and grandchildren, Jermicheal, La'Nia, Jalen, Bentley, and Bryliann, Mum, and Grandma MiMi love you all more than words could ever express. Thank all of you for sticking with me from the many miles traveled after completing my evening classes three times during the week. Now that we have completed this journey, you all have my word to be more intentional with my time of laughter and joy with each of you.

To my dad, Jeremiah, I want to thank you for being nothing short of the best dad a girl could ask for, although, you passed when I was 15, the lessons and time I had with you, my mum, and brother will never be forgotten. It is my prayer daily, although you all have passed away that I have made you, Mum (Brenda Faye), grandma (Georgia B.) grandma (Jessie Mae), and aunt (Aunt Linda Faye) proud!

#### ACKNOWLEDGEMENTS

Our Father God, you are truly magnificent and worth all the praise! There are a lot of loved ones, colleagues, mentors, and friends that I am indebted forever. I would be remiss if I did not thank my awesome committee members, Dr. David Baker, Dr. Danita Bailey-Samples, Dr. Jasmine Drake, and Dr. Howard Henderson. Dr. David Baker, I want to thank you for standing here with me until I crossed the finish line even after your much deserved retirement. Dr. Danita Bailey-Samples, our Father God knew exactly what he was doing when he allowed me to meet you in 2015, and you have been nothing but a blessing in my life personally and professionally along with your awesome husband, Dr. Samples. Dr. Jasmine Drake, I will never forget your mentorship and how you so graciously took me under your wings, all the advice you have provided me, and the patience you have displayed throughout this journey. Dr. Howard Henderson, I truly appreciate your splendid feedback, your expertise, and your support to ensure this dream to obtain my doctorate degree became a reality, I will never forget it. Each of you was instrumental and played a significant role in ensuring I completed what I started years earlier. Lastly, a special thanks to Dr. Helen Taylor-Greene who was my professor during this journey and continues to be that guiding light, mother figure, and mentor that pushes and encourages me to fulfill the desires of my heart both in my personally and professionally pursuits.

### CHAPTER 1

#### **INTRODUCTION**

Historically, the idea of students of all backgrounds and walks of life receiving an equitable education has been a popular but false narrative in the United States. For example, although on May 17, 2019, the race and social justice community celebrated the 65th anniversary of *Brown v. Board of Education* (1954), arguably the most pivotal milestone in American education, school segregation continues to exist. Academic achievement data indicates that many African American children and youth are not the beneficiaries of changed educational practices. They continue to struggle to reap the benefits of the promises of the Brown decision (Anti-Defamation League, 2014).

Several researchers have examined how zero-tolerance policy and exclusionary discipline practices have created a pipeline between the school, juvenile, and criminal justice systems in the United States (Mowen & Brent, 2016; Cuellar & Markowitz, 2015; Maxine, 2018). In Texas for more than a decade school discipline served as a gateway to the criminal justice system in which one-third of all youth in a correctional facility had dropped out of school. Thus, many researchers noted that racial disparities in academic achievement and exclusionary school discipline have been persistent features of U.S. public schooling for decades (Curran, 2016; Gregory & Weinstein, 2008; Kinsler, 2011; Magnuson & Duncan, 2006). Texas Appleseed (2007) noted that more than 80 percent of Texas adult prison inmates are school dropouts with minority students disproportionately represented in exclusionary discipline practices. A consistent and much more openly discussed problem in American education is the academic performance gap between students of color and White students (Paige & Witty, 2010). African Americans and other

minorities have not experienced as much progress as their White counterparts, which is reflected by urban high school graduation rates and rates of disciplinary actions (Musu-Gillette et al., 2016). For instance, nationally the graduation rate for Black students decreased from 13 to 6 percent, and the White student graduation rate decreased from 7 to 5 percent (de Brey et al., 2019). In Texas, Black students comprise about13% of the elementary school population and represented 47% of all pre-K through 5th-grade out-of-school suspensions (Craven, 2017). Consequently, a school-to-prison pipeline is possible with negative consequences for African American students' education and their communities (Kim, Losen, & Hewitt, 2010; Payne & Welch, 2010).

Do policy and practices in the educational and criminal justice systems work in a manner that results in a disproportionate number of African American students who are pushed out of school and into the prison system? Sixty-five years after *Brown v. Board of Education* (1954), the promise of equal access to quality education remains unfulfilled. Expulsion and suspensions in schools are one of the best predictors of who will drop out of high school, with African American students that are three times more likely to experience suspension or expulsion than their White peers. Moreover, Black students have a higher likelihood of experiencing both suspension and expulsion during their educational trajectory, than their White or Asian peers (Kewel et al., 2007). While *Brown vs. Board of Education* (1954) attempted to provide equitable access to schools for students of color, today it is often schools that act as gatekeepers to the successful matriculation of African American students. Zero-tolerance policy and exclusionary discipline practices have created structural inequality in the education sector, resulting in a decrease of African American students graduating and increasing the likelihood that

African American students will come into contact with the prison system. In the 1970s, suspension for minority students, more specifically, African American students, began to rise drastically. In 1975, The Children's Defense Fund published School Suspension: Are They Helping Children? a report that questioned disparities and the increase in in-school suspension and out-of-school suspensions (Skiba & Losen, 2016). Data from state and federal entities note a creditable trend of disproportionality in school discipline as it relates to African American students (Kim, Losen, & Hewitt, 2010; Skiba et al., 2011; Viadero, 2020). Smith (2009) described the school-to-prison pipeline as a prevalent trend of pushing marginalized students out-of-school and funneling them into the criminal justice system. For instance, youth who experienced a referral to juvenile probation from schools in 2015 included 25% Black, 53% Latino, 21% White, and 1% classified as others (Texas Appleseed and Texans Care for Children, 2016). Cardichon and Hammond (2019) and Texas Appleseed (2016) also noted that Black students comprised 40% of students suspended, however, they consist of only 16% of all public-school students with continued research and evidence illustrating that this is not because of worse behavior, but because of harsher treatment for minor offenses such as tardiness and disrespect, which has exacerbated racial disparities in Texas Public Schools.

The zero-tolerance policy is used to ensure that consequences are applied for all offenses regardless of how minor and or taken into consideration the rationale for the behavior. Zero tolerance became prevalent during the Reagan-Bush presidential years as the "war on drugs" became the nation's central focus and federal drug policy took a gettough stance to raise awareness that certain behaviors would not be permissible. More specifically, the zero-tolerance policy became educational legislation through the efforts of Democratic Senator Diane Feinstein in the early '90s and ultimately involved the Safe Schools Act and the Gun Free Act under President Clinton in 1993 (US S 854: Gun-Free Schools Act, 1993). The zero-tolerance policy was established as a policy that included expelling students who were found in possession of firearms on a school campus. Although it became evident that the zero-tolerance policy was ineffective as it relates to the "war on drugs," the concept proliferated in public schools across the nation. Research has illustrated no significant rise in school violence in the late '80s, even though school districts throughout the country adopted zero tolerance to include infractions including weapons, drugs, gangs, classroom disruptions, smoking, and dress code violations (Skiba & Noam, 2001). In 1994, the federal government mandated a zero-tolerance policy nationally after the signing of President Clinton's Gun-Free Schools Act. The mandate included an expulsion to be one academic school year for possession of a weapon or a referral of students who violated the law to the criminal justice system (Cohen, 2016). The policy became the umbrella for many other discipline infractions that consisted of suspending and expelling students for absenteeism, disruptive classroom behavior, defiance, and fighting (Skiba et al., 2002; Mendez & Knoff, 2003; Reyes, 2006; Stahl, 2016).

A zero-tolerance policy that results in the use of exclusionary discipline practices that consisted of suspensions (in-school suspension or out-of-school suspension), expulsions, and district alternative education placement (DAEP) that removed students from the learning environment has lifelong implications on a student's trajectory. Research also suggests that students who are suspended or expelled suffer academically and are more likely to drop out and become involved in the criminal justice system later in life. For instance, The Children's Defense Fund's report released in 1975, Suspensions: Are They Helping Children first brought the issue of racial disparities in discipline to national attention. Skiba and Williams (2014) noted that African American overrepresentation in out-of-school suspensions has progressively increased from the 1973 Office for Civil Rights data collection and estimates from the most recent release of the U.S. Department of Education's Office for Civil Rights that African Americans are about 3.5 times as likely to be suspended as White students. For instance, zero-tolerance policy and exclusionary discipline practices have been related to lower academic achievement (Beck & Muschkin, 2012; Mendez & Knoff et al., 2003; Skiba & Rausch, 2004), school dropout (American Academy of Pediatrics, 2013; American Psychological Association, 2008; Ekstrom et al., 1986), and involvement in the juvenile justice system (American Academy of Pediatrics, 2013; Balfanz et al., 2003; Fabelo et al., 2011; Nicholson-Crotty et al., 2009). Texas Education Agency Data for the 2018-2019 academic year noted that African American students accounted for 230,887 of the student population and White students were 260,119. However, 201 African American students accounted for Juvenile Justice Alternative Education Program enrollees as compared to 101 of their White counterparts (TEA-PEIMS, 2018-2019). Rafa (2019) noted in 2015-16 that approximately 2.7 million students in K-12 received one or more out-of-school suspensions, and over 120,000 students were expelled with or without educational services. This data, coupled with research illustrates substantial disparities in the application of suspension and expulsion based on race, gender, and disability. Meanwhile, in every state, Black students are more frequently pulled out of class, suspended, and/or violently arrested for vague and capricious reasons like "talking back"

or "having an attitude," or are punished for what they look like or wear. Not only do these administrative actions rob students of their right to an education, but they also put them on a pathway to the criminal justice system. Black people account for 13% of the U.S. population, but they represent close to 40% of the combined state and federal prison population (Carson, 2020), and once Black youth have been pushed out-of-school several times for various discipline reasons, they often encounter the prison system where they become a part of more dismal statistics.

To make equal opportunity a reality, we must push our school districts and teacher education programs to ensure that all school employees and educators unpack their own biases and learn ways to build safe school communities rooted in restorative justice rather than harsh punishment and exclusion (Stevens & Evans, 2018). According to Townsend-Walker and Townsend (2014), racial disparities in suspension have resulted in discriminatory policies and practices in public schools and halted progress towards the fair, equal treatment of students that should have been a result of the Brown decision. Where can Black students feel safe, when the school system, prison system, and other entities view them as a potential threat? There is a distressing sociopolitical context that might contribute to the school-to-prison pipeline that results in discriminatory discipline practices, that in turn, have a direct connection to pushing students out-of-school into the criminal justice system.

#### **Purpose of the Study**

This study explored factors that perpetuate the school-to-prison pipeline in a new outlook that contributes by providing an analysis of the implications of discipline practices within a Texas school district. Additionally, this analysis determined whether and to what extent discipline and academic gaps adversely impact African American students within a Texas school district. Discipline practices are not new, although the national movement towards a zero-tolerance approach has negative implications for African American students and institutions (Morgan et al., 2014; Bland & Harwin, 2017). Bland and Harwin (2017) recently reported that data from the Department of Education's Civil Rights Database found a significant increase in racial disparities in arrests and referrals to police following the institution of zero-tolerance policies.

Scholars and advocacy leaders have responded to disciplinary practices and the school-to-prison pipeline. However, critical pedagogy research that examines race and disproportionate minority contact for juveniles within the criminal justice system is limited. The research questions that guide this study are designed to increase our understanding of the pipeline and how it shapes the life course of African American students.

#### **Statement of the Problem**

Zero-tolerance policy and exclusionary discipline practices are not effective in promoting safer schools and have resulted in overzealous punishments that have caused more harm than good (Blackburn, 2016; Kang-Brown et al., 2013). Diliberti et al. (2019) conducted a recent report on crime, violence, discipline, and safety in U.S. Public Schools and found that, while discipline practices have become more punitive, they have not resulted in decreasing crimes or creating safer schools. Harsh punishments such as exclusionary discipline practices increase the likelihood of students dropping out and fuel the school-to-prison pipeline (Diliberti et al., 2019). For instance, Kang-Brown et al. (2013) pointed out that the zero-tolerance policy was created to discourage violence and gun possession, however, Johnson and Williams (2020) found that only 5 percent of disciplinary actions in recent years involved the possession of a weapon. The increase in crime has been attributed mostly to Texas's most vulnerable students, who are unequally punished for disruptive, and unruly behavior (Johnson Williams, 2020).

Types of exclusion include in-school suspension (ISS), out-of-school suspension (OSS), district education alternative placement (DEAP), and expulsions that result in the removal of students from instructional settings. From 2017 to 2018, black students made up 13% of the student population in Texas but received 33% of all out-of-school suspensions (Johnson & Williams, 2020). Additionally, it should be acknowledged that a school referral could potentially result in a student going from the classroom to the justice system as well as increase the likelihood of the student dropping out of school (U.S. Department of Education, 2015; American Psychological Association, 2008; Kang-Brown et al., 2013).

The history of zero-tolerance policy discipline in Texas dates back to 1992 when the State Board of Education implemented a zero-tolerance policy designed to address drug and school violence (Johnson, Wilson, & Green, 2015). Since then, in Texas, African Americans have been more likely than other students to be disciplined during their seventh to twelfth-grade years (Fabelo et al., 2011). In Texas, data was collected as it relates to several types of disciplinary actions including in-school suspension (ISS), out-of-school suspension (OSS), district alternative education placement (DEAP), Juvenile Justice Alternative Education Program (JJAEP), and expulsion, all of which illustrated disproportionality (Johnson, 2016). County-level data illustrates that Texas led the U.S. in days lost to suspension: 882,360 days of instruction were lost to out-of-school suspension per 100 students in 2015-16 (Austin, 2018).

Texas Appleseed (2016) and Craven (2017) acknowledged that Chapter 37 of the Texas Education Code mandates when students are removed from their classrooms for discipline violations, and the offenses, which require removal must include assault, drug use, weapons violations, and other offenses that threaten student or staff safety. Texas Appleseed (2016) and Craven (2017) acknowledged that the Department of Education found that although African American children were 18% of the national pre-school population, they accounted for 48% of the suspensions. In Houston ISD, 70% of out-of-school suspensions given to Kindergarten through 2nd graders were issued to African American boys (Craven, 2017).

Suspensions and expulsions are considered a key touchpoint in the school-toprison pipeline because students who experience expulsion have a greater likelihood of encountering the criminal justice system. Heitzeg (2016) indicated that the government is failing youth, especially the youth of color in schools as a result of a zero-tolerance policy as the policy has caused more harm and resulted in more juveniles becoming a product of the juvenile and criminal justice system. Mansoor (2017) conducted a question and answer with Dr. Reyes from the University of Houston about school discipline and the criminal justice system. It was noted that thirty-five percent of black men younger than 35 who did not complete high school are likely to go to prison, and high school dropouts are 20 times more likely than college graduates to be in prison. Thus, African American students tend to have a greater likelihood of encountering either the juvenile or adult criminal justice system. Pearman et al. (2019) claimed that there is growing interest in the relationship between the racial achievement gap and the racial discipline gap. This research examined the relationship between exclusionary discipline and a student's race and gender within a Texas public school district for students in 3<sup>rd</sup>-12<sup>th</sup> grade.

#### Significance of the Study

Racial disparities in school discipline and the juvenile justice system in Texas resemble national trends that appear to value exclusionary discipline practices over student achievement. Even though African American students comprise a small percentage of students, their negative experiences often undermine their chances of achieving the American Dream. Research indicates that although African American students are disciplined more often and more severely, the infractions are no greater than those of their White counterparts. Craven (2017) acknowledged that these biases have caused educators in some cases to punish African American students more harshly and more frequently than others for the same behaviors. The problem is finding solutions to address the disparities and prevent the pipeline. The significance of this study is to encourage policymakers to consider revising and or replacing the existing zero-tolerance and discipline practices which may essentially benefit all students regardless of their race, especially when it relates to closing the discipline and achievement gap for African American students. Thus, the study provides a greater understanding and direction for future research into the implications of exclusionary discipline practices and zerotolerance policy as it relates to African American students' achievement and discipline disparities in a Texas school district.

This study is timely due to the recent increase of widely publicized confrontations between African American students and White adults in their respective classroom settings. The school-to-prison pipeline has been characterized as a disciplinary system that criminalizes students, exacerbates academic gaps, leads to students' exit from school, and increases the likelihood of students' involvement in the juvenile-adult criminal justice system. Existing evidence also suggests that the Black-White gap in high school graduation is approximately 8 to 9 percentage points for those born in the 1980s (Heckman & LaFontaine, 2010; Murnane, 2013). Closing high school graduation gaps across different racial and ethnic groups are of concern to researchers and policymakers because attainment disparities carry significant implications for African American students' future economic well-being (Tamborini et al., 2015).

Since the introduction of the zero-tolerance policy and exclusionary discipline practices, Texas school districts, like thousands of districts across the nation, have implemented these practices and policies without a thorough analysis of the implications. This quantitative study encourages policymakers to assess the zero-tolerance policy to include discipline practices in schools as well as contribute to the growing body of literature by attempting to comprehend and offer reasonable recommendations concerning exclusionary discipline practice's impact on the graduation rates of African American students.

In addition, the study explored the implications of the zero-tolerance policy and attempts to move from current research that focuses on the individuals as independent agents that are essentially responsible for the size of the school-to-prison pipeline. Instead, this examination will delve into the school-to-prison pipeline as an interconnected and highly developed system in which African American students become trapped. Further, the study offers policy implications that could provide school districts with an evaluation of current discipline policy trends for the assessment of discipline effectiveness in their respective districts. The introduction of the zero-tolerance policy and exclusionary discipline practices in Texas school districts, like many thousands of districts across the nation, has not provided enough information about how these policies impact students and educators in specific locales.

#### **Research Questions**

For this study, the examination of the relationship between exclusionary discipline and the school-to-prison pipeline was guided by the following research questions:

- RQ 1: Is there a significant relationship between student race (Black or White) and exclusionary disciplinary practices within a Texas public school district?
- RQ 2: Is there a significant relationship between student gender (Male or Female) and exclusionary disciplinary practices within a Texas public school district?
- RQ 3: Is there a significant relationship between student race (Black or White), student gender (Male or Female), and exclusionary discipline practices within a Texas public school district?

#### **Statement of Hypotheses**

The following research hypotheses were formulated for the present investigation:

H1: There is a significant relationship between student race and the exclusionary discipline practices of the school district in the study.

- H2: There is a significant relationship between student gender and the exclusionary discipline practices of the school district in the study.
- H3a: Controlling for gender, race affects the exclusionary discipline practices of the school district in the study.
- H3b: Controlling for race, gender affects the exclusionary discipline practices of the school district in the study.

#### **Operational Definitions of Key Terms**

The following variables and terms are operationally defined to provide clarity and understanding relative to the focus of the present research.

- <u>Achievement Gap:</u> The achievement gap refers to the disparity in academic performance between minority and low-income students and their peers. (Rowley & Wright, 2011). Academic performance is measured by standardized-test scores, course selection, and high school graduation rates, often highlighted with substantial performance gaps between Black and Latino students, at the lower end of the scale, and their White peers. In addition to racial disparities, there are similar academic disparities between students from low-income families and students from higher-income families (Paige & Witty, 2010; Nasir, 2012).
- <u>Critical Race Theory</u>: An academic discipline focused on a critical examination of society and culture, through the intersection of race, law, and power (Alexander, 2010; Ladson-Billings, 1998). "Race is a social construction that exists for separation and stratification of the dominant group" (Crichlow, 2015).

- 3. <u>Disciplinary Alternative Education Programs (DAEP)</u>: The Disciplinary Alternative Education Program (DAEP) provides an educational placement for students who have been removed from the traditional school setting due to weapons, drugs, acts of violence, or at the administrator's discretion for repeated disruptive behavior, disrespect, truancy, etc. in the regular school setting (Booker & Mitchell, 2011).
- <u>Discipline Gap</u>: The discipline gap refers to minority students and students with disabilities who receive discipline referrals and harsher consequences at a disproportionate rate when compared to their White peers (Skiba et al., 2011; Townsend, 2000; Welch, & Payne, 2010).
- <u>Exclusionary Discipline:</u> "Creating practices that remove students from instruction such as suspensions and expulsions" (Texas Association of School Board, 2019).
- <u>Expulsion</u>: banning a student for at least one calendar year (Gagnon et al., 2016).
- Pushout: School pushout refers to harsh discipline policies that push students out of school with excessive out-of-school suspension and expulsion (Gregory, Skiba, & Noguera, 2010).
- 8. <u>School to Prison Pipeline</u>: The "school-to-prison pipeline" refers to the policies and practices that push students out of schools and into juvenile and criminal justice systems (Kim, Losen, & Hewitt, 2010). Factors involved with the school-to-prison pipeline include inadequate access to quality schools, disparate discipline practices, disciplinary alternative school settings, and

criminal justice system involvement that may act as barriers to returning to traditional public schools (Alexander, 2010; Fowler, 2011; Kim, Losen, & Hewitt, 2010; Smith, 2009).

- 9. Suspension (In-School Suspension-ISS & Out- of School Suspension-OSS: The Texas Education Code (TEC) Chapter 37 outlines the circumstances under which a student may be removed from the classroom. TEC, Section 37.002 authorizes a principal to place a student in ISS or OSS, another disciplinary setting when removed from the classroom (Texas Association of School Board, 2019).
- 10. <u>Zero Tolerance:</u> Zero Tolerance policies were originally introduced to school districts as a means of ensuring safe campuses as it relates to weapons, drugs, and violent acts on school grounds (Skiba & Peterson, 2000). Over time, however, zero tolerance has come to refer to school or district-wide discipline policies that mandate predetermined, often harsh punishments (such as suspension and expulsion) for a wide range of school policy violations that go beyond weapons, drugs, and violence (Noguera, 2003; Skiba & Peterson, 2000).

#### **Organization of the Study**

This dissertation is structured to introduce the study and illustrate the effects of discipline practices on African American students. Chapter 1 presents the development of the phenomenon of the academic performance gap between African American students and their White counterparts as it relates to the school-to-prison pipeline, and how disciplinary practices have permeated the American-Texas school system. The chapter

provides historical and contemporary information that is essential to understanding the intersectionality of race, disciplinary practices, and the school-to-prison pipeline. Chapter 2 provides a review of relevant disciplinary research and the theoretical framework that guides the study. It includes prior studies of the relationship between disciplinary practices and educational outcomes for African American students. Chapter two concludes with a summary of research about how the prison system has infiltrated the school in the form of policies, such as zero-tolerance policy and exclusionary practices. Chapter 3 presents the research methodology and includes the research questions, operational definitions of study variables, description of the sources of data, statistical analyses, as well as the strengths and limitations of the study. The dataset includes school achievement, school discipline, as well as the life outcomes for African American students in a Texas school district. Chapter 4 presents the study findings. This dissertation concludes with chapter 5 summarizing the study findings, and recommendations for policy, practice, and future research.

#### CHAPTER 2

#### LITERATURE REVIEW

Historically, the educational experience for African American students has been one of marginalization and racially fueled policies and practices that paved the way for the current educational status of African American students. The past several decades provide evidence of the potential negative implications of exclusionary discipline practices as they relate to African American students' achievement gaps, which have garnered increased attention from educational leaders, policymakers, and researchers (Losen & Martinez, 2013; Skiba et al., 2011; Steinberg & Lacoe, 2017). Additionally, there is a preponderance of evidence that exclusionary discipline approaches are used disproportionately across student race and gender (Blake et al., 2011; Brown & Tillio, 2013; Curran, 2016; Losen & Martinez, 2013; Losen & Skiba, 2010; Pearman et al., The Children's Defense Fund (1975) published one of the earliest reports 2019). documenting the overrepresentation of Black students' in-school suspensions. Since then, multiple studies have reported similar results. For example, Rocque (2010) found evidence of a Black-White disciplinary gap in disciplinary referrals, and Costenbalder and Markson (1998), Gregory and Weinstein (2008), and Skiba et al. (2014) reported racial disparities concerning in and out of school suspensions.

This quantitative research study examined the intersection of race and disciplinary practices as it relates to the school-to-prison pipeline and the implications for African American students in a Texas public school. There is limited research as it relates to the correlation of race, critical pedagogy, and disproportionate minority contact for juveniles within the criminal justice system.

For instance, many scholars suggested how there is a critical need for more research on effective approaches to close academic gaps and readdress discipline practices that reify racism in schools and classrooms (Gregory et al., 2016). The present research study will incorporate the "zero tolerance policy" and the exclusionary disciplinary practices as they relate specifically to African American students. The present investigation will explore the demographics of African American students who experience exclusionary discipline and how this impacts their ability to succeed academically. Next, the school-to-prison pipeline demographic trends concerning the use of suspension and African American students who may have been "pushed-out" will be addressed as part of this study. Finally, the limitations of the extant research and the need for further investigation will be analyzed. Many studies have been developed to understand the rationale behind why students are not matriculating through high school graduation. The primary goal of this study was to raise awareness as it relates to the revisiting-eradicating zero-tolerance policy, and exclusionary disciplinary practices. In conclusion, this study will focus on a Texas Public School District and its possible implications as they relate to the criminal justice system.

#### **Historical Background**

Research on school discipline practices, including exclusionary discipline, dates back to the early 1970s (Morgan et al., 2014). Nearly 40 years after the zero-tolerance mandates, Ladson- Billings (1998) noted that inequalities in educational attainment, school punishment, and discipline (e.g., zero-tolerance policy) are pipelining African American students to what has been referred to as the school-to-prison pipeline. Curtis (2014) and Busby, Lambert, and Lalongo (2013) criticized zero-tolerance policies for

their failure to curb violence in schools, the disproportionately adverse effects on minority students, and how the policy contributes to increased rates of recidivism. Despite being a small percentage of many school districts' populations' African American students accounted for the largest demographic experiencing exclusionary discipline for relatively minor infractions as well as the achievement gap (Kim, Losen, & Hewitt, 2010). Furthermore, the Anti-Defamation League (2014) and The U.S. Department of Education Civil Rights Division (2014) noted. that Black students are suspended and expelled at a rate three times greater than that of White students. Since the introduction of the zero-tolerance policy and exclusionary discipline practices, Texas school districts, like thousands of districts across the nation, have implemented these practices and policies without a thorough analysis of their implications. For instance, low academic achievements or unsupportive environments may lead to disciplinary referrals and push students out-of-school, which increases their likelihood of not embracing the pathway to graduating high school, attending college, and/or trade schools, but increases their chances of coming in contact with the criminal justice system.

#### **Theoretical Framework**

Two theories were utilized in this investigation Critical Race Theory and Critical Pedagogy as it relates to the academic success of African American students in graduating high school. In addition, to critically examine the research, it is imperative to do so from the lens of analyzing policies in school and analyzing disciplinary gaps through the lens of Critical Race Theory and Critical Pedagogy. Operationalizing the research questions and analyzing data through two theoretical frameworks, will assist with the structure of the study.

#### **Critical Race Theory**

Conceptualized by scholars and theorists Bell (1976) and Delgado (1989) Critical Race Theory (CRT) was developed in the 1980s by scholars of color who responded to critical legal studies and civil rights scholarship. Critical Race Theory, no longer restricted to the legal domain for scholars, encompasses a broad range of researchers and fields (Rabaka, 2006). DeCuir and Dixon (2004) affirmed CRT as a method of qualitative analysis when investigating the experiences of Black students in a predominantly White school. Ladson-Billings (1994) indicates that CRT is grounded in eradicating all forms of racism, racial subordination, and discrimination. Delgado and Stefancic (2001) posited that CRT emphasizes the socially constructed nature of race and asserts judicial conclusions to be the result of the workings of social phenomena but perceives race as an essential factor. As it relates to Critical Race Theory, Ladson-Billings and Tate "proposed that Critical Race Theory (CRT) is a framework that was developed by legal scholars, however, could be employed to examine the role of race and racism in education" (Dixson & Rousseau, 2005, p. 8), so this further informed the current use of Critical Race Theory with the research. The tenant of Critical Race Theory that describes racism as pervasive (Ladson-Billings, 1999) was the underpinning of the study. CRT regards racism as so deeply embedded in society that it is often overlooked and viewed as ordinary (Delgado, 2009).

Critical Race Theory has been applied to determine the history of how race and racism have manifested with the Pre- K-12 pipeline, and most importantly, the work has enabled individuals to engage in often difficult challenges within the classroom as it relates to the context of the zero-tolerance policy as well as the community work

(Ledesma & Calderón, 2015). Additionally, Critical Race Theory offered tools that would allow one to engage these concerns in the framework of policy as well as highlight the prevalence of racism throughout education (Ledesma & Calderón, 2015). The tenets of the Critical Race Theory framework note that race should be the center of discussions as it relates to equity and justice. The tenants used for this analysis are 1) the intersectionality of race and racism with gender, class, sexuality, and other forms of subordination in maintaining educational inequality; 2) the challenges of the dominant ideology regarding culture and intelligence, language and capability, meritocracy and objectivity; 3) the commitment to social justice and the Freirean notion of critical consciousness; and 4) the utilization of any interdisciplinary approaches that analyze and articulate the linkage between social inequality and schooling (Solorzano & Yosso, 2002). Critical Race Theory informed the research because race, along with student gender, was a variable when determining the impact as it relates to exclusionary discipline. Delgado and Stefancic (2001) noted CRT regards racism as being deeply embedded in society that is often overlooked and viewed as ordinary and ingrained within the systems of American society.

Additionally, policies and practices may not necessarily target members of a race, but if they disproportionately impact members of a given race; they are viewed as functions of institutionalized racism. In addition, Critical Race Theory informed the research as it relates to analyzing policy in Pre-K-12<sup>th</sup> that impacts communities of color in disparaging ways. CRT was used to critique curricular practices, tracking procedures, teacher expectations, and intelligence testing (Solorzano & Yosso, 2002). Schools have policies and practices in place that are embedded in racism. However, it seems Critical Race Theory has become an increasingly permanent solution for educational researchers aspiring to critically examine educational opportunities, school climate, representation, and pedagogy. For CRT to continue being fruitful as it relates to shifting the paradigm, it is recommended that enriched dialogues continue that will equalize the educational opportunities for students of color (Ledesma & Calderon, 2015). Institutional racism functions in a manner that is often considered subtle in contrast to the blatant bigotry of the past (Alexander, 2010). Therefore, Gilborne (2006) and Ladson-Billings (1999) asserted that racism is illustrated by the outcome of practices, and not evident by intent. Fundamentally, policies and practices may not have targeted members of a certain race; however, if they disproportionately impact members of a certain race, this is viewed as a function of institutionalized racism.

Through the lens of Critical Race Theory this could be powerful as one investigates the current state of public education today, 66 years after Brown, when schools are more segregated than ever before. Moreover, as we know from experience, a policy is fundamentally needed when it pertains to influencing the school climate; so, Critical Race Theory informed the research to analyze the impact of policies regarding Pre-K-12 education, especially when it relates to examining the impact of policies like zero-tolerance-policy. It should be noted that critical pedagogy has its roots within critical theory. One of the central figures that contributed to critical pedagogy is the Brazilian educationalist Paulo Freire. He contributed to both its theoretical and practical development. One of his significant contributions was a discussion about critical consciousness, *conscientizacao*, as consciousness beyond understanding which leads to action (Freire, 1970).

#### **Critical Pedagogy**

A major factor of retention is student engagement in the classroom, which can be impacted by course design and instruction. As it relates to Black and Brown students, who experience exclusionary discipline at a higher rate than their White counterparts, previous research shows that the likelihood of suspensions for Black students declines when they are taught by a large population of Black teachers. Critical pedagogy aims to value all students' experiences through six elements of course design and instruction: decreasing teacher power, student self-reflection, dialogue, student voice, critical analysis, and action. Critical pedagogy is a theory regarding classroom practice. Critical education theorists have argued that: "School knowledge should have a more emancipatory goal than churning out workers...School knowledge should help create the conditions productive for student self-determination in the larger society that can only be achieved when class society is abolished" (McLaren, 2015, p. 211).

Critical Pedagogy notes the importance of students from marginalized communities having the ability to connect with teachers to be challenged about social inequalities that impact as well as enable them to become the agents of change in their respective communities. Critical Pedagogy informed the research because it notes the importance of marginalized communities-students successfully having the opportunity to engage in a learning environment that will enable them to become agents of change versus participants within special education assignments and exclusionary discipline consequences (Waitoller & King, 2016; Welsh & Little., 2018).

Ladson-Billings (1999) remarked critical pedagogy is concerned with transforming the power of education for individuals and society for the purpose to create a more equitable and just society for everyone. Critical Pedagogy informed the research because it noted how reimagining the relationships between teachers, students, and the established knowledge that is being taught in the classroom could increase students' engagement and lessen defiance in the classroom. Moreover, Critical Pedagogy informed the research because when students are disengaged and defiant in the classroom, this results in referrals that essentially increase their risk of entering the school-to-prison pipeline (Rivers, 2020). Moreover, McLaren defines critical pedagogy as a way of thinking about, negotiating, and transforming the relationship between classroom teaching, the production of knowledge, the institutional structure of the school, and the social and material relations of the wider community, society, and nation-state (McLaren, 2014).

Additionally, Critical Pedagogy offers teachers practices and guidelines for countering existing conditions and achieving social transformation as critical agents of change. Student discipline is unique from academic achievement insofar as disciplinary referrals are about teachers' perception of student behavior and are thus strongly subject to the teachers' conscious and unconscious biases (Milner, 2013). Discipline referrals are as much, if not more grounded in teachers' and other school personnel's perspectives of the severity of the student's misbehavior as they are in the objective severity of the student's behaviors (Annamma et al., 2019). This will essentially allow students to participate in their learning as opposed to ultimately being problematic and later sent to the office for disruptive behavior, which often has consequences that include exclusionary discipline.

Fundamentally, students should have the full freedom to question and assert one's voice; this freedom is central to the purpose of education. Critical Pedagogy informed the research because, as Paula Freire noted, education can serve as a vehicle for social and economic transformation, which, as noted previously when students are sent to the office, they are missing invaluable classroom instruction. In addition to missing invaluable instructional hours, this increases their chances of experiencing exclusionary discipline, which also increases their chances of entering the school-to-prison pipeline (Jemal, 2017).

Critical Pedagogy further informed the research because the pioneer Paul Freire firmly believed that educational change must be accompanied by vital changes in the social and political structures in which education takes place. In the case at hand, it is vital to understand that the zero-tolerance policy that was designed to address violations such as bringing a firearm to school is now used to justify sending students to the office with referrals for being defiant. After so many defiant referrals a student will receive some form of exclusionary discipline, which increases the student's chances of entering the school-to-prison pipeline. Giroux (2013) recognized that Paulo Freire's work regarding critical pedagogy identifies the conditions in which students learn how to read and write as well as learn and master how to incorporate the basic concepts of literature, the arts, science, philosophy, social theory, and the applied disciplines.

Critical pedagogy offers and informs the research as it relates to teaching practices and guidelines for schools' maintenance of the status quo and for achieving social transformation in solidarity with historically marginalized communities. Frequently, Black children are negatively impacted in schools because of school staff that typically have the power to label, classify, and define, who often do not always have marginalized children's best interests at heart. Foiles noted White teachers internalized racist beliefs they relate to Black and Latinx children. Moreover, researchers concluded that internalized racist beliefs held by White teachers are a primary contributor to the disproportionate rate at which Black and Brown students are disciplined as compared to their White counterparts (Hancock & Warren, 2017). In compassion to their White counterparts, Black students experienced the harshest and most exclusionary forms of school discipline (Carter, 2018: Gregory et al., 2010).

By using Critical Race Theory and Critical Pedagogy lenses to inform the research, this analysis focuses on the student experience as well as noting a historical and contextual account that analyzes the educational experiences and data trends of African American students in a Texas school district. Many students who have behavioral issues are acting out in response to stressful or unsafe conditions that may exist in their respective homes and/ or neighborhoods. Therefore, issuing exclusionary discipline such as OSS places them in the very environment(s) that may be problematic or unsupervised, and may cause more harm. For example, while the students are serving suspension or expulsion, they are more likely to engage with other students that may be removed for the same if not similar offense or engaged with individuals that may have already come in contact with the criminal justice system. With the application of the theory and teaching philosophy, the researcher will note institutional factors that impact African Americans being pushed out of public schools and essentially pipelined to the criminal justice system. The relationship between race can be explored through CRT and critical pedagogy (CP).

## **Race and Zero Tolerance Policies**

# **History of Zero Tolerance Policies**

Zero-tolerance policies as defined in the previous chapter were instituted in American public schools over 20 years ago to address discipline reforms in the 1990s. Since then, zero-tolerance policies have generated research, criticism, and discussion about their effectiveness in improving discipline, they are doing very little to deter violence in schools, and their adverse impact on minority students (Busby, Lambert & Lalongo 2013; Curtis, 2014). Bouchein (2015) and TABSE (2019) acknowledged the negative implications of zero-tolerance policies for public-school students, especially African American and special education students. Riddle and Sinclair (2019) found biased administrators or local voters often use their sociopolitical power to support policies (e.g., zero-tolerance policies or random drug sweeps) that disproportionately punish students of color. Zero tolerance has created what we have come to know as the pipeline between schools and the criminal justice system.

Zero tolerance policies adversely affect African American learners in school settings nationally. The use of this practice has been noted as the underlying cause of minority students and special education students being funneled through the criminal justice system. The Gun-Free Schools Act of 1994 embodied a framework that mandated states to expel students who bring firearms to school. Although initially intended as a response to serious offenses (e.g., selling drugs or engaging in gang-related fights on school grounds) to ensure safety at schools, in recent years zero-tolerance policies have been applied for minor offenses (e.g., being disrespectful or talking back (Flannery, 2015). Bouchein (2015) pointed out that numerous schools expanded their use of zero-

tolerance policies not only to include what was mandated by the law but also to apply this to other school infractions (e.g., student violence and drugs). Steinberg and Lacoe (2017) pointed out that the initial zero-tolerance policies were to remove disruptive students so that school campuses' climate, as well as safety concerns, would be addressed to allow teachers and students to have a safe learning environment. However, research has noted that the effect was otherwise. For instance, it was noted that middle school campuses expelled Black students at four times the rate of White students and Latino students at two times the rate (Loveless, 2017). The Civil Rights Data Collection Report for the 2015-2016 school year noted that African American students represented approximately 15% of the total students in the U.S. public schools; however, African Americans were 31% of those referred to law enforcement or arrested (CRDC, 2016; TABSE, 2019).

Steinberg and Lacoe (2017) revealed that students at campuses with high suspension rates reported feeling less safe than those on campuses with similar students' lower suspension rates. Steinberg and Lacoe (2017) avowed that research from the Association Zero Tolerance Task Force determined from 10 years of research that exclusionary discipline policies have not reduced school violence and have not had the wavering results of ensuring schools are safer.

Contrary to the negative implications as they relate to African American students and ZTP, some researchers believe ZTP is fulfilling the desired results just as it did when implemented in the 1990s. Ujifusa (2018) and Vara- Orta (2018) expressed that Betsy Devos, the former Secretary of Education believes rescinding the 2014 Obama Discipline Guidance will make it clear that the classroom teachers and local school leaders will have the autonomy to discipline as they believe necessary without fearing costly civil rights investigations.

## Zero Tolerance Policies and Discriminatory Practice

As previously mentioned, the zero-tolerance policy is a systemic contributor to the sequence of events that many describe as the school-to-prison pipeline. In fact, many of the provisions noted under the zero-tolerance policies warrant suspension and expulsion of students who are found to have violated the policies, which often enable school districts to pursue criminal or juvenile justice involvement. Zero-tolerance policies were derived from the nationwide war on drugs. By the late 1990s, zero-tolerance policies emerged in the school system, with an intended focus on deterring the possession of weapons on campus. In addition, the myth of a class of genetic "super predators" can be said to have originated in 1965. The term "super predator," which was coined by Princeton University professor John DiIulio (Bouchein, 2015; Boghani, 2015), evolved as a label for young Black males, Dilulio also claimed in 1995 and 1996 that America was facing a looming threat from a population of young Black boys who were "Godless" and "alien" and who can "kill or maim on impulse without any intelligible motive" (Bouchein, 2015; Boghani, 2015).

With a perceived notion that youth were becoming increasingly violent, many legislators concurred that something must be done; therefore, the Gun-Free Schools Act of 1994 was signed and passed into law (Cerrone, 1999). The passages mandated that all schools adopt zero-tolerance policies for infractions involving weapons present on school campuses. For instance, it was clearly articulated that school districts receiving Title 1 funding must adhere to the guidelines, with one of the non-negotiable guidelines being expulsion for a minimum of one year as well as a referral to the juvenile or criminal justice system if a student was found in possession of a gun (Bouchein, 2015; Curtis, 2014).

Correspondingly, the adoption of the zero-tolerance policy by schools roughly began in the early 1990s with media overrepresentation and dramatization of youth violence. For instance, it was noted that media overrepresentation was a pressing concern that created an image of youth as "superpredators" (Bouchein, 2015; Boghani, 2015). According to Cardichon and Hammond (2019), exclusionary discipline because of zero-tolerance policies adopted at the local and state level was initially intended to deter students from engaging in violent or illegal behavior because of the consequences. Additionally, stringent zero-tolerance policies have been extended and applied to nonviolent offenses as well as most subjective offenses (American Psychologist, 2008).

On the contrary, Ujifusa (2018) avowed that President Trump created the School Safety Committee because of the Parkland Florida school incident in which the campus allegedly concealed many discipline infractions to prevent Civil Rights Investigations, and as a result, the shooter was given a Federal Commission on School safety report. Thus, the Trump administration officially rescinded the Obama-Era School Discipline as a direct result of the Parkland Florida shooting, with many civil rights advocates noting this was unrelated to the issue of exclusionary discipline and students of color (Rafa, 2019). It was further confirmed that superintendents feared they would lose federal funds in the pressure to adopt discipline policies that appear to undercut teacher authority and allow parents to note their perspectives as they relate to school discipline and school safety (Eden, 2018).

Balingit (2018) recognized that teachers and administrators have been unfairly blamed for the discipline disparities, whereas Eden (2018) asserted that the disparities are evidence of a much larger problem, such as a student's socioeconomic status and whether they reside in a two-parent household. Eden (2018) further acknowledged that, after the rescinding of Obama Discipline Guidance 2014 by the Trump administration, parents, teachers, and local school boards will be able to pursue sensible discipline policies.

Miller (2019) insisted a major reason for the rescinding of the Obama Discipline Guidelines that were implemented in 2014 was the desire to address the racial disparities that existed because of zero-tolerance policies that ultimately increased the use of exclusionary discipline practices. Additionally, Miller (2019) disclosed that the primary reason for the change was the result of researchers arguing that disparities in discipline existed between Black and White students. Moreover, many have observed the disproportionate application of school discipline and punitive policies to students of color as a step toward the cradle-to-prison pipeline (Bristol & Mentor, 2018: Britton, 2021a; Britton, 20201b).

### **History of Zero Tolerance Policies and Implemented**

#### **Practices to Eradicate the Use of this Practice**

Pearman et al. (2019) maintained that the Civil Rights Data clearly illustrated discipline gaps and highlighted the adverse impact zero-tolerance policies have had as they relate to African American students and the criminalization of this student group. There is a considerable amount of research indicating that zero-tolerance policies and the use of exclusionary discipline for nonviolent behavior are ineffective in changing student behavior and creating a safe learning space for all students (Pearman et al., 2019).

According to Curtis (2014) and Bouchein (2015), during the academic school year 1996-1997, 79% of public schools had implemented zero-tolerance policies to address student violence. Additionally, 94% of policies or applications of those policies involved student possession of firearms, 91% for weapons pertaining to items other than guns, and 88% for drugs (Bouchein, 2015; Curtis, 2014). The study also revealed that after the implementation of zero tolerance this measure was also used for discretionary offenses such as insubordination, truancy, and disrupting class (Fowler et al., 2011). Consequently, as zero-tolerance policies were used in a manner that did not distinguish between nonviolent and violent offenses, the American Bar Association noted that schools somehow gained the ability to discipline and punish more students harshly (Bouchein, 2015).

Instead, Kamenetz (2018) suggested that the question is whether the government's decision to rescind Obama Discipline Guidance would bring back the days of zero tolerance. Correspondingly, Harper et al. (2018) performed an analysis for NPR of the federal data by Child Trends that illustrated suspension declined with the Obama Discipline Guidance of 2014, but Black high school students were still twice as likely to be suspended nationally. In the same way, many individuals fear that without federal government intervention there may be little to no pressure for change in states such as Texas, which experienced an increase each year as it relates to suspension from 2012 to 2016 (Harper, 2019).

Visibly, discipline practices have been a topic of research dating back to the early 1970s. However, the national movement is also known as the Superpredator scare, which involved criminologists in the 1990s predicting a coming wave of "superpredators."

Many researchers believe this frightening imagery was racially coded (Equal Justice Initiative, 2014), so it is vital to explore this new phenomenon in education. In the same way, Cardichon and Hammond, (Equal Justice Initiative, 2014) upheld that zero policies that apply strong punishments often consist of removing a student from campus with the consequence of suspension or expulsion. Thus, this prediction of students as young as those in elementary school carrying guns, as opposed to lunches, may have exacerbated the issue drastically, as an increase of African American students experienced more consequences associated with the exclusionary discipline (Kamenetz, 2018; Bornstein & Miller, 2019).

#### **Race and Exclusionary Discipline**

Among the conspicuous unforeseen consequences of the use of exclusionary discipline practice is the notion that this practice mirrors the school-to-prison pipeline. Bland and Harwin (2017) reported data from the Department of Education's Civil Rights Database that illustrates a significant increase in racial disparities for Black student arrests and referrals to the police. Similarly, de Brey et al. (2019) recorded that Black children represented 15 percent of all students during the 2015–16 academic school year, while they represented 31 percent of students arrested or referred to police for their behavior on school campuses. Many other studies are illuminating the detrimental effects of school discipline policies and practices as they relate to African American students. Steinberg and Lacoe (2017) discerned those racial disparities in suspension begin as early as preschool, with Black students comprising 18% of enrollees in preschool, and they represent 48% of preschool children receiving one or more suspensions.

Comparably, Massar et al. (2015) realized out-of-school suspensions are used more often than any other consequence for various reasons in public schools within the United States. According to Nowick (2018) and Cardichon and Hammond (2019), it has been documented in the past three decades that African American students, especially African American males, are overrepresented among those subjected to exclusionary discipline, particularly, suspension and expulsion. For instance, it has been verified that children of poverty, as well as students with academic issues, are overrepresented as it relates to exclusionary discipline.

Equally, Steinberg and Lacoe (2017) asserted that Black and Hispanic students are more likely than White students to receive suspension or expulsion for minor offenses such as inappropriate verbal language. For example, the study illustrated that county-level estimates of racial bias were evident in data from approximately 1.6 million visitors to the Project Implicit website, which noted racial disciplinary disparities across approximately 96,000 schools in the United States covering around 32 million White and black students (Riddle & Sinclair, 2019).

Similarly, the U.S Department of Education Office for Civil Rights (2018) discovered that during the 2013–2014 school year, Black students from pre- k to 12<sup>th</sup> were 3.8 times as likely than White students to have one or more out-of-school suspensions. The data showed that among students from Pre-K to 12<sup>th</sup>, 6% received one or more out-of-school suspensions, however, the percentage differed by race and gender with 18% for Black boys, 10% for Black girls, 5% for White boys, and 2% for White girls. Furthermore, the data revealed that Black children constituted 19% of preschool

enrollment; however, they consisted of 47% of children receiving one or more out-ofschool suspensions.

Likewise, the research reflected that the use of zero tolerance with strong punishment is overutilized because many districts and schools are applying these policies to nonviolent offenses and subjective offenses such as talking, excessive tardiness, or truancy. Cardichon and Hammond (2019) and Bouchein (2015) corroborated that after the implementation of zero-tolerance policies the suspension rate increased nationwide with 3.5 million students being suspended within the academic school year. Specifically, in 1974 1.7 million students were suspended annually from public schools in the United States, and by 2011-2013 that number had doubled, resulting in almost 3.5 million students (Bouchein, 2015).

Bouchein (2015) conducted a study comparing disciplinary approaches as they relate to zero-tolerance policies, which appear to be a direct reflection of exclusionary discipline in the Maryland and Texas Public Education Systems. Bouchein (2015) selected the two states of Maryland and Texas due to the states' recent revision of their discipline policies as well as the fact that both states have a diverse student population. It was discovered that suspension rates dropped for both states for all races; however, the racial disparities have increased and remained unchanged in both states. For instance, Black students' risk of facing exclusionary discipline compared to their White counterparts increased in Maryland from 2.4 to 3.0 and in Texas from 4.3 to 4.9 (Bouchein, 2015). The study found that Maryland and Texas implemented policies that reduced suspension rates but that continued disparities among Black students.

Conversely, Ujifusa (2018) noted AASA, the School of Superintendents surveyed 950 school districts leaders and found that only 16% of school districts had modified their discipline practices because of the 2014 Obama Discipline Guidance with less than 1% of the respondents noting a negative impact on school personal ability to administer discipline while 7% percent noted a positive impact. Vara-Orta (2018) claimed Betsy Devos, former Secretary of Education believed that every student has a right to attend a school that does not discriminate and one that does not treat them as statistics. However, she believed that the Obama discipline guidelines led to negative implications for campuses. For instance, she indicated that teachers and advocates felt the discipline guidance resulted in decisions based on a student's race, and statistics became more important than students' and teachers' safety (Vara- Orta, 2018).

Epstein (2014) agrees that the zero-tolerance policy may have led to punishments that fail to fit the violation and believes that it may be fair to ask if the zero-tolerance policy is logical in an educational context. Epstein (2014) further mentioned that the Department of Education may have erred in arguing against the policy merely based on disparate impact. Epstein (2014) concluded that zero-tolerance policies often prove conducive when the school setting is clear about the policies. For instance, if the policy embedded suspension for a student who is found in possession of a gun or drugs this allows students to note the consequences and encourages them to behave appropriately.

Additionally, Bouchein's (2015) study also predicted that one in three students from kindergarten through 12<sup>th</sup> grade would experience some type of exclusionary discipline. Bouchein (2015) found that when suspension was used, and the infraction did not involve a weapon (e.g., insubordination), the suspension tripled the student contact

with the juvenile justice system. In other words, when exclusionary discipline such as inschool suspension and out-of-school suspension was used for minor infractions, this action increased the students' chances of entering the juvenile justice system. Losen et al. (2015) deduced that with the increased use of exclusionary discipline 18 million days of instruction were lost annually.

## **Discriminatory Practices as They Relate to AA students**

Research regarding exclusionary discipline more specifically, suspensions among African American students have consistently found negative and unintended consequences for this subgroup. Actually, TABSE (2019) posited that African American students had a high attendance rate; however, their learning was compromised as it relates to ISS and OSS. For example, the student count for out-of-school suspension was 402,373; however, African American students consisted for 32% (131,474) of this number for the 2016-2017 academic school year compared to14% (57,765) for White students. TABSE (2019) also asserted that younger African American learners' experience with exclusionary discipline was worse. In the United States, African American students represented 18% of preschool student enrollment; however, 48% of this age group experienced out-of-school suspension more than once. In Texas, African American children represented 14.9% of the total preschool student enrollment and accounted for 37.8% of students who received more than one out-of-school suspension. Similarly, Losen et al. (2015) disclosed that students of color are suspended from school for minor offenses that do not pose a serious threat to school safety, with several studies noting African American students receive suspensions for less serious violations in comparison to their White peers.

On July 3, 2018, the Trump administration chose to rescind the guidance despite the research and studies illustrating the effectiveness of these efforts. Gillette et al. (2018) noted that efforts to make schools safer based on the implementation of the Obama administration series of guidance proved conducive to the National Center for Education Statistics' Indicators of School Crime. Additionally, the 2017 survey illustrated that the nation's schools are becoming safer, with a reduction from 85-89% between 1999-2000 and 200-2010, which compares to a 2015 survey that notes a 79% reduction (Gillette et al., 2018).

Alternatively, Cardichon and Hammond (2019) reported on December 21, 2018, that the Trump administration rescinded this guidance despite a substantial body of research illustrating that zero-tolerance policies and exclusionary discipline practices for nonviolent behavior are largely ineffective as they relate to changing student behavior and creating safe learning spaces. Additionally, Cardichon and Hammond (2019) confirmed that there has been much research on discriminatory discipline practices and their negative implications as they relate to students of color. Cardichon and Hammond (2019) and TABSE (2019) also substantiated that African American youth are far more likely than other students to be suspended and expelled. For instance, African American males in Texas represent 6% of students enrolled in public school, but they represented 30% of students who were suspended. This is compared to White male Texas students who represented 24% of students enrolled in public school, but they represented 16% of those who were suspended.

In the same way, TABSE (2019) data collected from the Civil Rights Data Collection clearly illustrates a problem as it relates to exclusionary discipline and the criminalization of African American students, which has resulted in the promise of education for this population becoming one that has no promise and is broken. Massar et al. (2015) conducted an evaluation of 1,840 public middle schools in the United States and found that exclusionary discipline practices continue to be used in response to student misbehavior. However, they found that suspension is unlikely to change the students' behavior. Their study discovered that 6.6% of students who received a suspension in a school year were 71.9% more likely to receive another suspension throughout the school year (Massar et al., 2015).

On the contrary, Vara-Orta (2018) asserted that the Federal Commission on School Safety reported from a survey of teachers that the 2014 Obama Discipline Guidance coerced teachers to lower their discipline to non-exclusionary methods that in some cases were most likely not in accordance with the infraction that the student committed. In addition, the faculty also recognized a significant increase in safety concerns for student and teacher safety (Vara-Orta, 2018).

Conversely, Miller (2019) reported findings from previous research conducted by lead researcher Paul Wright that suggest the use of suspensions by teachers and administrators may not have been biased, as some scholars suggest. Furthermore, the study accesses students of color and their families' lack of academic success and focuses on shifting Black students' behavior as the solution versus suggesting a shift in structures of policies that have systematically failed students of color (Miller, 2019).

# Race and Pushout (Graduation and Dropout rates; Critical Race Theory; DMC) School-to-Prison Pipeline History

Historically, students engaged in delinquency and were forcibly removed from the educational setting because of exclusionary discipline practices (e.g., the suspension was typically left unsupervised while the student was not in school, increasing their chances of involvement in deviant and criminal behavior during the suspension) (Losen & Martinez, 2013). Additionally, there is much research conducted on the negative behavioral implications, but there is also literature regarding exclusionary discipline and its negative implications as they relate to negative academic outcomes for African American students.

Obviously, arrests in school represent the most direct routine that pipelines students from school-to-prison. With systematic factors of zero-tolerance policies and exclusionary discipline, minority youth are at higher risk of coming in contact with the school-to-prison pipeline. It should be stressed that the "School-to-Prison Pipeline" (STPP) refers to the overrepresentation of minority students, particularly African American males, in the juvenile corrections system and, consequently, in the prison system (Children's Defense Fund, 2009; Losen & Martinez, 2013; Skiba, 2011). Furthermore, Black students in the United States are subject to disciplinary action at rates much higher than their White counterparts.

## **Race and Disproportionate Minority Contact (DMC)**

The Office of Juvenile Justice and Delinquency Prevention (OJJDP) defines DMC as an overrepresentation of minority youth who come in contact with the juvenile justice system (Slowikowski, 2009). Bouchein (2015) asserted that scholars recognized that the

initiation of the zero-tolerance policies in public school discipline policies such as (ISS, OSS, DAEP, and expulsions) have created the funneling of disadvantaged students into the juvenile and adult criminal justice system, more commonly known as, the school-to-prison pipeline. Bouchein (2015) and the American Civil Liberties Union (2019) stated that some scholars suggest the school-to-prison pipeline is "facilitated through many other factors such as police presence on school campuses, providing officers with more power to discipline students, criminalizing minor code of conduct infractions, failing schools that are highly segregated by race and income, poor resources, and the adoption of zero-tolerance policies" (ACLU, 2019; Heitzeg, 2009). Moreover, according to many scholars, the adoption and implementation of zero-tolerance policies have been cited as the primary factor in the furtherance of the school-to-prison pipeline.

Notwithstanding all the research concluding the negative implications, there is only a small proportion of states and cities that have deviated from the use of zerotolerance policies to address student misbehavior (Bouchein, 2015). According to Bouchein (2015), schools with lowered suspension rates have better test scores, compared to those that have an alarming amount of use of suspension. Besse and Capatosto (2018) have shown that we must work to address the racial disparities evidenced across a variety of educational outcomes including academic achievement, school climate, and punitive discipline policies that have been cited as contributing factors to pushing students of color into the criminal justice system. Within the student population, a small number are most at risk of being captured within what has been ubiquitously called the "school-to-prison pipeline," sometimes targeted by authority figures, and subject to recidivism.

Often discipline rules are established by school districts (e.g., student codes of conduct) which do not allow flexibility as it relates to discipline alternatives outside of suspensions and expulsions (Steinberg & Lacoe, 2017). For instance, during the 2015-16 school year, African American students represented 15% of student enrollment, and enforcement represented 31% of referred students law to were arrested. Correspondingly, across the five largest racial and ethnic groups in 2016-17, grades 7-12 dropout rate was highest for African American students (2.1%), followed by Hispanic (1.7%), multiracial (1.0%), White (0.8%), and Asian (0.4%) students (Steinberg & Lacoe, 2017).

Similarly, news headlines and videos posted online have reinforced the story told by the statistics. For example, a black high school student in South Carolina was thrown from her desk by a school resource officer for refusing to put away her phone (Ford, 2016); a 12-year-old Latina girl in Texas was body-slammed and nearly knocked unconscious by police following a verbal altercation with another student (Bever, 2016); a black middle-school boy in Virginia was arrested for allegedly "stealing" a free carton of milk (Wise, 2016). These stories have assisted in raising public awareness of the negative experience African Americans students encounter as a result of exclusionary practices and zero-tolerance policies.

Ford (2016) submitted that this problem is not new. However, since the early 1970s overlapping with the start of widespread desegregation efforts, the racial gap in suspensions has been trending upward. This trend has been caused in part by the adoption of zero-tolerance initiatives that demand heavy-handed approaches to the slightest disciplinary infractions. Zero tolerance has not proven effective as a preventative measure; instead, it has contributed to increased truancy, dropout rates, and encounters with law enforcement (Harper, 2008; Skiba, 2001). Also, according to Steinberg and Lacoe (2017) students removed from school due to exclusionary discipline practices often have lower achievement on standardized exams and are more likely to repeat a grade, drop out of school, and eventually, encounter the juvenile justice system.

According to the U.S. Department of Education Office for Civil Rights (2018), in Texas, African American students represented approximately 12.7% of students enrolled, but 20% of students were referred to law enforcement and/or arrested. TABSE (2019) highlighted that these practices have ushered the criminal justice system into the school system causing the distinction between African American and African American criminals to become blurred. Additionally, TABSE (2019) proposed that African American learners are gravely impacted by the unforgiving legal system, and if not adequately addressed there will be a continued increase of African American students introduced to the pipeline to prison rather than the progress from cradle to college.

Admittedly, some scholars have even advanced the idea that such disparities are evidence of a "school-to-prison pipeline" that targets disadvantaged and minority youth (Wald & Losen, 2003). African American students are often faced with the risk of being held back, dropping out, or ending up in the criminal justice system. Epstien (2014) reported a New York Times argument that zero-tolerance policies have resulted in serious and lasting consequences for students who are suspended or expelled. According to Balingit (2018), Black students accounted for 15% of the student body during the 2015-2016 school year but accounted for 31% of arrests. Balingiit (2018) further declared two

years earlier that Black students accounted for 16% of the student body and 27% of school arrests.

## **Gender and Exclusionary Disciplinary Practices**

School discipline literature has not given much attention to Black females compared with Black males. Studies on Black female discipline experiences have mostly examined discipline sanctions of Black boys compared to Black girls, with Black girls rarely being mentioned outside of descriptive statistics (Skiba et al., 2002). Instead of demonstrating how disproportionate discipline practices might also have detrimental effects on Black girls' school experiences, the literature appears to report Black females' discipline sanctions as a way to draw attention to the disparity between Black males' and Black females' discipline experiences. Beginning in elementary school and lasting through high school, disproportionate discipline appears to be a problem for Black girls. Black girls have a higher percentage of suspensions than Hispanic or White girls (Raffaele Mendez et al., 2002). A study published by Raffaele Mendez and Knoff (2003) replicated the research of Taylor and Foster (1986) and demonstrated that suspension rates for Black girls were higher than those for White and Hispanic girls in primary and secondary schools. The types of behavior infractions that result in disproportionate discipline of Black girls are not well understood due to the lack of research on the discipline experiences of girls.

Mixed results have been found regarding race/gender dynamics. Studies have found that African American males are suspended more frequently than other race/gender groups (Skiba et al., 2002), but this is not the only demographic experiencing race disparities (Annamma et al., 2019). Blake et al. (2011), for instance, found that African American females were twice as likely to be suspended as Latinas and non-Latina White females. The categories of infractions also varied across races and ethnicities. Black women in that study were suspended for acting in ways that deviated from traditional female behaviors (Blake et al., 2011).

The purpose of a study by Blake et al. (2011) was to expand the school discipline literature by investigating whether Black girls in schools are disciplined disproportionately. Specifically, Blake et al. (2011) investigated whether African American girls experienced greater discipline infractions and sanctions than White and Hispanic girls. The study assessed female students from elementary and secondary schools in an urban school district in the Midwest with at least one discipline sanction. Data were gathered from a school-record database that identified 38 possible discipline sanctions ranging from a warning to expulsion. An examination of exclusionary discipline sanctions, such as in-school and out-of-school suspensions, was reported in the study. In all discipline sanctions, Black females were disproportionately affected. Researchers found that Black girls were twice as likely to be suspended in school and out of school as their same-gender peers based on exclusionary discipline practices. Comparing the results, the study revealed that Black female students received twice as many in-school suspensions as White female students and nearly four times as many outof-school suspensions as White female students. Generally, Black females are involved in the school discipline system at rates similar to Black males. The Blake et al. (2011) study found that girls of color were more likely to receive exclusionary discipline sanctions and were twice as likely to be suspended at school and outside of school. Black girls,

therefore, have a much higher probability of experiencing exclusionary discipline than White girls.

Approximately double the suspension rate for Hispanic males (22%) and less than a quarter of the suspension rate for African American males (42%) can be found in national data from 1993 (National Center for Education Statistics, 2016). African American males in grades 6 through 12 were suspended at similar rates in 2012 when almost half of them, 48%, had been suspended. This was more than double the rate for non-Hispanic White males (21%) and Hispanic males (23%). Additionally, female students have reported race-based disparities. Female African American students enrolled in grades 6<sup>th</sup> through 12<sup>th</sup> had been suspended 29% more times than both White non-Hispanic and Hispanic students enrolled in grades 6<sup>th</sup> through 12<sup>th</sup>.

### **Intersection Between Race and Gender**

In a study of the intersection of race and gender, Haight, Kayama, and Gibson (2016) found that Black students with family support- involvement decreased their likelihood of experiencing exclusionary discipline. In 2016, Haight et al., criticized school-to-prison pipeline research for failing to account for Black girls' intersection of race and gender experiences. In 2014, Irby outlined several strategies to enforce discipline through consistency in rules enforcement, consistency in expectations, and the use of preset sanctions. The trust students have in rules and practices, according to Irby (2014), will motivate them to follow them. Studying the intersection of gender, race, and discipline contributes to the growing literature on how the school-to-prison pipeline impacts young people of color.

Black girls are statistically more likely to be suspended or expelled, according to Crenshaw, Nanda, and Ocen (2016). According to the researchers, policymakers should evaluate and revise policies that push girls into the juvenile justice system as well as develop best practices that ensure school personnel enforces each student's right to be free from sexual harassment and bullying. Racism is a factor that influences disproportionate disciplinary practices according to Bryan et al. (2012). In order to address disparities in school discipline, Bryan et al. (2012) suggest that systemic and gap-reducing interventions be examined.

A study conducted by Yale University researchers Gilliam et al. (2016) used data found in the National Prekindergarten Study (NPS). The study analyzed expulsion rates by school setting, gender, and race/ethnicity in public schools, Head Start, and private providers. The researchers found racial disparities in education in 2016 and identified several factors, including bias, discriminatory practices, school racial climates, and insufficient professional development for teachers. According to the study on implicit bias, racial, gender, and exclusionary discipline recommendations including suspension and expulsion are interconnected. Due to the lack of research on the interaction between gender and race disparities in suspensions, Gilliam et al. (2016) recommend that gender and race disparities in suspensions be explored.

## Race and Push-Out (Critical Pedagogy)

## History of Push-Out (Critical Pedagogy) as it relates to AA students

The consistent racially biased and ineffective removal from the educational setting supports the funneling of numerous children of color from schools into the juvenile justice system. It has been well documented that Black students are not in the classroom as often as their counterparts, and this is a result of discipline and further hinders their access to quality education. McIntosh et al. (2010) confirmed that the most negative discipline encounters originated in the classroom, and the referral procedure is often initiated in the student's classroom, with a teacher who often has not developed a positive rapport with African Americans students. For example, in some instances, teacher-issued discipline referrals resulted in students being sent to an administrator where they were typically assigned consequences such as in-school or out-of-school suspension (McIntosh et al., 2010). Although more research should be conducted to confirm the achievement gap and disciplinary gap for students in certain school districts, this analysis is timely due to the recent increase of widely publicized confrontations between African American students and White adults in their respective classroom settings.

Bouchein (2015) conducted a study involving a key component of zero tolerance that focused on removing disruptive students from the learning environment, and how this resulted in schools increasing their use of in and out-of-school suspension for both violent and minor infractions. Cardichon and Hammond's (2019) research has demonstrated that zero-tolerance policies have resulted in negative consequences for students to include attainment, academic achievement, and a student's welfare. In addition, the report further indicated that states other than Texas have adopted less punitive approaches and reduced the use of exclusionary practices to ensure a more inclusive learning environment for all students.

Musu-Gillette et al. (2018) confirmed that the National Center for Education Statistics' Indicators of School and Crime and Safety 2017 survey showed that push-out is when a student is noted as at-risk. In this report, a student identified as at risk of dropping out of school is one who is under age 26 and who meets the following criteria: has been placed in an alternative education program under TEC §37.006 during the preceding or current school year.

Fortunately, many researchers, such as Skiba, Losen, and Blake, have applied an empirical analysis to the data. In Closing the School Discipline Gap (Losen et al., 2015) these researchers and others offer a more nuanced look at the discipline disparity phenomenon, bringing a few things to light. For example, various studies have found that students of color are more likely to be reprimanded for subjective offenses not specified by the school (insubordination, disrespect, excessive noise, and so on) and based on a judgment call of a teacher or administrator.

In contrast, White students' punishments are more likely to be for objective offenses for which the school requires a categorical sanction (drugs, weapons, obscene language, and so on). Students of color, black students, in particular, are more likely than White students to be referred to the office or suspended, even when the misbehavior is similar. This is not just a disproportionate representation; it is differential treatment by the system.

TABSE (2019) divulged that the Civil Rights Data Collection illustrates the disproportionate representation of African American students who are excluded from the instructional learning environment and placed in alternative environments that often do not support cultural competencies or academic excellence. Moreover, it was indicated that racial disparities exist at the elementary and secondary levels. Additionally, TABSE (2019) noted that African Americans are undoubtedly being pushed out of public schools

through forms of exclusionary discipline, which often are ISS and OSS, for infractions categorized as defiance of authority.

Researchers have concluded that students may not be willing to assimilate into cultural norms and practices that are foreign, unknown, or uncomfortable to them. So, rather than suspending the students, teachers should be trained in cultural competence so that African American students can transition from one culture to another one without compromising their identity.

Correspondingly, Jimenz and Flores (2019) clarified that the impact of exclusionary discipline practice varies. However, students suspended and expelled lose valuable instructional time, feel less connected to school, and tend more frequently to drop out. Vara- Orta (2018) examined the Obama Discipline Guidelines that were implemented in 2014 to adequately address the pushing out of students of color. For instance, the guidelines urged school leaders to seek other alternative suspensions and other consequences that would remove students from the classroom because Black and Hispanic students were suspended at a much higher rate in comparison to their counterparts. The suspension is correlated to increased dropout rates as well as lower academic achievements.

Vara-Orta (2018) reported that the National Association of School Psychologists guideline was vital in deterring systemic disparities, implicit biases, and discipline policies for minor and subjective offenses including insubordination or disrespect. Kamenetz (2018) identified a growing body of research that illustrates that being suspended, expelled, or arrested at school is correlated with a higher dropout rate as well as lifelong negative implications. In the same way, Kristen Harper, director of Child Trends, exclaimed that one suspension can make a difference, and statistics illustrate a negative impact on students of color that is disproportionate to their actual behavior.

In the same way, Massar et al. (2015) conducted a brief examination of the use of suspensions in 1,840 public schools in the United States and noted that suspension is associated with a higher likelihood of academic failure, school dropout, and involvement in the juvenile justice system. According to Judith Browne Dianis, executive director of the national office of the Advancement Project, the data concerning the overrepresentation of Blacks in exclusionary discipline clearly illustrates racism is alive in our American school system and illustrates as well that Black students are less safe and more restrained and pushed out of school more than other student groups. Kaitom Nammer, an attorney with the Advancement Project, explained that studies noted that Black students have no discernible differences in the way they behave compared to their counterparts. She believes the disparities exist because of the way adults are responding to students' behavior.

Additionally, as noted by Bouchein (2015), a single suspension has negative implications as it relates to students' long term. Bouchein suggested that one suspension or expulsion doubles the likelihood that a student will repeat a grade and this experience has been one of the strongest predictors of students dropping out of school. Also, according to Bouchein (2015), when schools see an increase in police presence, students are also arrested and referred more often to juvenile justice (Curtis, 2014 (Curtis, 2014).

Based on Steinberg and Lacoe's (2017) research, as of May 2015, 22 states and the District of Columbia revised their laws to encourage schools to limit their use of exclusionary discipline practices, implement support that is non-punitive and provide counseling support as well as dropout prevention for students at risk. Moreover, during the 2015-2016 school year, 23 of the 100 largest school districts nationwide implemented policy reforms that mandated non-punitive discipline strategies or the limited use of suspensions.

#### **Summary of the Research Literature**

Diverse student populations have increased, prompting a shift towards educational policies that move beyond curriculum and instruction and improve academic outcomes in learning institutions (Johnson, 2002). Disproportionality could occur due to a variety of factors, including behavioral differences, classroom referral bias, or disciplinary prejudices. Research on zero-tolerance policies and exclusionary discipline impact on African American students has illustrated no academic benefits in the use of these policies (Gordon, 2018). The policies and practices are strongly associated with low achievement, a heightened risk of dropping out, and a greater likelihood of criminal justice involvement (Miner & Blake, 2018). Research has not yielded the alignment between disciplinary reforms recommended at higher levels of governance, such as the state, and policies implemented by leadership at the local school district level. For example, a study of zero-tolerance policies found that school districts expanded the list of offenses to which exclusionary policies applied beyond those required by state law (Pearman et al., 2019). Disciplinary disparities have been documented but revisiting or eradicating exclusionary practices and policies because of the implications for African American students has not been fully examined. Furthermore, this issue has not been addressed since the advent of zero-tolerance policies and exclusionary discipline practices in the 1990s. As we know, education is the key to success.

Moreover, it could be argued that suspension from school places at-risk students at a higher risk for academic failure because it limits their connection to teachers and positive educational outcomes. Students who experience exclusionary discipline suffer a greater likelihood of suffering academically, dropping out, and engaging in delinquent behavior (Skiba et al., 2014). As Skiba et al. (2014) pointed out, the American Psychological Association evaluated zero-tolerance policies and concluded that such policies do not make schools safer. Additionally, the study declared that such practices are disproportionately applied to students of color and those with disabilities, and the consequences are not developmentally appropriate for children and adolescents (Skiba et al., 2012).

Schools suspend and expel students at a disproportionate rate based on race. The ostensible purpose of exclusionary discipline is to prevent the student from dropping out of school or entering a juvenile or adult facility Due to statistical data, which shows that African American males are more likely to be suspended than other population groups, prior research on the school to prison pipeline focused on African American males. The majority of pipeline studies fail to consider African American girls. This study provides a comprehensive review of empirical evidence and the need for consistent reviews of school-wide discipline plans and the effectiveness of revisiting or eradicating exclusionary school discipline practices and policies on the state and federal levels.

### **CHAPTER 3**

## **DESIGN OF THE STUDY**

The primary purpose of this study was to determine if exclusionary discipline practices and policies in the educational and criminal justice system work in a manner that adversely impacts Black students (race or gender), and to what magnitude in a Texas public school district. The phenomena that are investigated in this examination are the intersection of zero-tolerance policies, exclusionary discipline practices, race, gender, and the implications for African American students within a Texas Public School District. Chapter 3 includes the type of design, data source, data collection, identification of dependent and independent variables, data analysis, and summary.

## **Type of Design**

A quantitative, ex post facto design was used because the numerical data collected was gathered from a public archival data source. Basler (2012) stated, "Ex post facto is a Latin phrase meaning 'from after the fact' and relies on observation of relationships among phenomena as they occur naturally without intervention from the researcher" (p. 49). The ex post facto design allows for the comparison of two or more groups of individuals with similar backgrounds who were exposed to different conditions as a result of their natural histories. This type of research will be the most appropriate design for this study because both the dependent and the independent variables have already occurred (Gay et al., 2016).

Moreover, through secondary analysis of data from a Texas Public School District, the researcher measured the institutional and individual mechanisms that disproportionately pull and push students of color into the "school-to-prison pipeline." The examination explored the predictors of school discipline contact and the resulting consequences of encountering this discipline. The examination proceeded to explore the relationship between school severity and various educational and juvenile justice outcomes. The "school-to-prison pipeline" (Wald & Losen, 2003) described an "increasingly punitive and isolating" path through the education system for African Americans and other at-risk students. Specifically, this chapter details the data source, data collection, identification of independent and dependent variables, data analysis, and summary.

# **Data Source**

The data source used to compile records for the population of interest was an archival set from a Texas Public School District PEIMS Department, which is data that must be submitted annually by all school districts to the Texas Education Agency. The Public Education Information Management System (PE1MS) databases served as the primary source of instrumentation for this analysis. The Texas Education Code §37.001 (2002) outlined the rules and procedures for enacting an exclusionary discipline consequence. In-school suspension is established under Texas Education Code §37.002 and is an action taken by an administrator that removes a student from his or her assigned classroom. Assignment of this consequence may not exceed 10 school days. Out-of-school suspension is established under Texas Education Code §37.005 and is an action taken by an administrator that temporarily removes a student from his or her home campus. Assignment to this consequence may not exceed 3 school days.

The Disciplinary Alternative Education Program (DAEP) is established under Texas Education Code §37.0051 and is an action taken by an administrator against a student who is temporarily removed and placed in a disciplinary alternative education program under Section 37.008, and the juvenile justice alternative education program under Section 37.011. The database encompasses all data received by TEA as it relates to public education, including student demographic and academic performance, personnel, financial, and other organizational information (Texas Education Agency, 2006).

## **Identification of Dependent and Independent Variables**

**Dependent variable.** In this study, the dependent variable was disciplinary actions for acts of disobedience and sanctions ranging in severity from ISS to Expulsion. Exclusionary discipline such as out-of-school suspension and expulsion is the focus of this analysis. The key variables consisted of African American ethnicity and disciplinary infractions. Disciplinary infraction was a key factor in this analysis. The entire analysis focused on the frequency and proportion of African American disciplinary infractions.

**Independent variable.** In this study, the independent variables were race and gender. The first variable is race, which was measured by the official record data of the participants who were identified as either White or African American. For this analysis, White and African American students were analyzed.

### **Data Collection Procedures**

The researcher accessed raw data regarding in-school suspension, out-of-school suspension, DAEP placement, and expulsion, all of which were secured via the secondary data retrieved from a Texas Public School District. The student information secured was in relation to his or her (race and gender, by student count), and assignment (with action to 1SS, OSS, DEAP, Expulsion)

# **Data Analysis**

For statistical purposes, the researcher used the applications of the Statistical Package for the Social Sciences (SPSS) to treat the data. Multiple regression analysis is appropriate for this study because it allows for examining a relationship between two or more variables and determining predictability among those variables. The hypotheses were tested at the .05 level of significance or better in this empirical investigation.

# Summary

The discourse in this chapter included the selected research methodology and the appropriateness of its design for the research study. The analysis should encourage future researchers to examine more closely school discipline disparities, especially as they relate to African American students. With an increased number of African American students being excluded and/or pushed out of school classroom instruction (i.e., as a result of exclusionary discipline for subjective offenses that are often viewed as minor offenses) further studies should be explored.

# CHAPTER 4 RESULTS

# Introduction

An underlying assumption of this study was that the zero-tolerance discipline policy disparately impacts students of color. The primary purpose of this study was to determine if exclusionary discipline practices and policies in the educational and criminal justice system work in a manner that adversely impacts Black students (race or gender), and to what magnitude in a Texas public school district. The study was guided by the following research questions:

- RQ 1: Is there a significant relationship between student race (Black and White) and exclusionary disciplinary practices within a Texas public school district.
- RQ 2: Is there a significant relationship between student gender (Male or Female) and exclusionary disciplinary practices within a Texas public school district?
- RQ 3: Is there a significant relationship between student race (Black and White), student gender (Male or Female), and exclusionary discipline practices within a Texas public school district.

This chapter summarizes the results of the analysis of the data. The statistical tests used in the analysis of the research questions included Chi-Square and multinomial logistic regression analysis. A descriptive analysis was performed to better understand the demographic characteristics of the study participants. Variables in the three research questions proposed by the study were tested. Chapter 4 is presented in the following sections: descriptive characteristics of participants, examination of hypotheses, analysis, and assumptions. Finally, the last section analyzed the three statistical (null) hypotheses formulated for the study. All three hypotheses were tested at the .05 level of significance or better.

# **Demographic Characteristics of Participants**

Participants in this study were students from grade 3 through grade 12, who received an exclusionary discipline (i.e., in-school or out-of-school suspension, expulsion, and DAEP) during the 2016-2017 school year. There were four hundred seventy (69.8%) African American and 248 (36.8%) White students for a total of 673 students. Regarding gender, there were 203 (30.2%) females and 470 (69.8%) males. See Table 1.

## Table 1

## Frequency Characteristics by Gender and Race

Variables	Frequency	Percent	
Female	203	30.2	
Male	470	69.8	
African American	425	63.2	
White	248	36.8	

Table 2 reflects discipline types and frequency. Discipline types were divided into three categories: DAEP, ISS, and OSS. There were 34 (5.1%) District Alternative Placements (DAEP), 466 (69.2%) In-School Suspensions (ISS), and 173 (25.7) Out-of-School Suspensions (OSS). See Table 2 for these results.

# Table 2

<b>D</b> '	•	1.	T
1 110	nn	11100	Inno
$\mathcal{D}$ is	civ	une	Type

Discipline Type	Frequency	Percent	<b>Cumulative Percent</b>
DAEP	34	5.1	5.1
ISS	466	69.2	69.2
OSS	173	25.7	25.7
Total	673		100.0

The results for exclusionary discipline practice by race are presented in the table below. There were 17 African American students representing 50 percent of all students placed in DAEP, while 17 White students representing 50 percent were the remaining students placed in DAEP. In contrast, 337 (72.3%) African Americans students received ISS, as opposed to 129 (27.7%) White students. Finally, 112 (64.7%) African Americans students were placed in OSS and 61 (35.3%) White students. See Table 3.

# Table 3

Number and	Percent of	Exclusionary	<i>Discipline</i>	<i>Practice by Race</i>
	<i>J</i>	~	1	-

Exclusionary Discipline	African Americans		Whites		Total	
Practice	n	%	n	%	n	%
DAEP	17	50	17	50	34	5.0
In School Suspension	337	72.3	129	27.7	466	69.3
Out of School Suspension	112	64.7	61	35.3	173	25.7

## **Examination of Hypotheses**

Ho<sub>1</sub>: There is no statistically significant relationship between student race

(Black or White) and exclusionary disciplinary practices.

Race was categorized into Black and White because this was the selected group of

students for this study. Each race was grouped into three discipline types. Blacks 17 or

50.0 percent were assigned to DAEP, of the total population, similarly, 17 Whites (50.0%) were also placed in DAEP. Placement for Blacks in ISS was 337 (72.3%), and 129 (27.7%) Whites. Furthermore, 112 (64.7%) of Blacks were put in OSS, as opposed to 61 (35.3%) of Whites.

## Table 4

Crosstabulation for Race versus Exclusionary Discipline Practices

Discipline Type* Race Crosstabulation								
	DA	DAEP ISS		OSS		Total		
	n	%	n	%	n	%	n	%
Black	17	50.0	337	72.3	112	64.7	425	100
White	17	50.0	129	27.7	61	35.3	248	100
Total	34	100.0	466	100.0	173	100.0	673	100

In Table 5 are the Chi-square test analysis results regarding race and exclusionary disciplinary practices. The value of the chi square statistic is 3.509, df = 4, p-value is (.477). Since the p-value is greater than the .05 alpha level, the Ho<sub>1</sub> hypothesis is accepted. Therefore, the data suggests that there is no statistically significant relationship between student race (Black or White) and exclusionary disciplinary practices. See Table 5.

# Table 5

Chi-square test for Race versus Exclusionary Discipline Practices

			Significance
	Value	df	(2-sided)
Pearson Chi-Square	3.509	4	.477
Likelihood Ratio	3.334	4	.504
N of Valid Cases	673		
Chi Sauara $-2500 df - 4 n - 477$			

Chi-Square = 3.509, df = 4, p = .477

Ho<sub>2</sub>: There is no statistically significant relationship between student gender and the exclusionary discipline practices of the school district in the study.

Regarding the variable exclusionary discipline by gender in DAEP, 1 or 0.1 percent were identified as female, and 33 (4.9%) were male. In contrast, 148 or 22.0 percent of females were identified as placed in ISS, while, 318 (47.3%) were male. On the contrary, there were 52 or 7.7 percent of females in OSS and 121 (18.0%) males. See Table 6.

#### Table 6

Number and Percent of Exclusionary Discipline Practice by Gender

	Female		Male		Т	otal
Exclusionary Discipline Practice	Ν	%	n	%	n	%
DAEP	1	0.5	33	7.0	34	5.1
In School Suspension	148	72.9	316	67.7	466	69.2
Out of School Suspension	54	26.6	121	25.3	173	25.7

The Cross-tabulation was computed to assess the count and expected relationship between the demographic factors of gender and exclusionary discipline practices. As revealed in Table 7, the actual count for females sent to DAEP was 1 and the expected was 10.3. In contrast, 33 males were assigned to DAEP, and the expected count was 23.7. The number of female occurrences in ISS was 148 and the expected count was 140.6. Ascribed for males ISS count was 318 and was expected to be 325.4. Regarding placement in OSS, females accounted for 54, and the expected count was 52.2. In contrast, the number of males counted in OSS was 119, whereas, the expected count was 10.8. See the results in Table 7 below.

# Table 7

			Disci	pline	
		DAEP	ISS	OSS	Total
	Count	1	148	54	203
<b>F</b> 1	Expected	10.3	140.6	52.2	203.0
Female	Count				
	Count	33	318	119	470
	Expected	23.7	325.4	120.8	470.0
Male	Count				
	Count	34	466	173	673
	Expected	34.0	466.0	173.0	673.0
Total	Count				

# Cross-tabulation for Gender versus Exclusionary Discipline Practices

Revealed in Table 8 are the Chi-square test analysis results regarding gender and exclusionary disciplinary practices. The value of the chi square statistic is 12.615, df = 2, p-value is .002. A significant relationship was found to exist between student gender and exclusionary discipline practices. Consequently, Ho<sub>2</sub> was rejected. See Table 8.

# Table 8

Chi-square test for Gender versus Discipline

			Significance
	Value	df	(2-sided)
Pearson Chi-Square	12.615	2	.002
Likelihood Ratio	17.717	2	.000
N of Valid Cases	673		
Chi-Square = $12.615$ , $df = 2$ , $p = .002$			

Ho<sub>3</sub>a: Controlling for gender, race does not affect the exclusionary discipline practices of the school district in the study.

A multinomial logistic regression analysis was conducted to assess whether race and gender had a significant effect on the likelihood of observing each response classification of exclusionary discipline practices. Multinomial logistic regression was performed to model the relationship between the predictors and membership in the three groups (race, gender, and exclusionary discipline), with the disciplinary infractions being the dependent variable. The traditional .05 criterion of statistical significance was employed for all tests. The addition of the predictors to a model that contained only the intercept significantly improved the fit between model and data, x2 (df=6, N=673=20.018, Nagelkerke R2 = .038 p = .003). McFadden's R-squared was calculated to assess the model fit, where values greater than .2 are indicative of models with an admirable fit (Louviere et al., 2000). The McFadden R-squared value calculated for this model was 0.02. As shown in Table 9, significant unique contributions were made by gender, but not by race. Therefore, Ho<sub>3</sub>a is accepted.

## Table 9

	Model Fitting Criteria	Likeliho	od Ratio T	ests
Model	-2 Log Likelihood	Chi-Square	df	Sig.
Intercept Only	69.222			
Final	Final 49.204		6	.003
	Pseudo R-Sq	uare		
Cox and Snell			.029	
Nagelkerke			.038	
McFadden			.020	

#### Model Fitting Information and Pseudo R-Square

Ho<sub>3</sub>b: Controlling for race, gender does not affect the exclusionary discipline practices of the school district in the study.

Indicated in Table 10 are the results of Likelihood Ratio Tests that assessed the goodness of fit of two competing statistical models based on the ratio of their likelihoods. Only one predictor had a significant parameter for comparing the ISS group with the DAEP group. The value of the chi-square statistic for gender is 16.684, df = 2, p-value is .000. For gender, the likelihood of a female being in the ISS group rather than the DEAP multiplicatively increased by 2.695. One of the predictors had significant parameters for comparing the OSS group with the DAEP group. The likelihood of being in the OSS group rather than the DAEP group was multiplicatively increased by 2.721 for females. Thus, hypothesis Ho<sub>3</sub>b: was rejected. The multinomial logistic regression analysis was used because it would allow the researcher to see ISS, OSS, and DAEP simultaneously to determine if a student's race or gender and any impact and if so to what magnitude it relates to this particular school district. This statistical analysis was selected because the exclusionary discipline is comprised of ISS, OSS, DAEP, and expulsion, hence, the statistical analysis would be used to determine from the view of all exclusionary disciplines being tested simultaneously to determine what impact the students' race and gender would have on the categories of ISS, OSS, DAEP, and expulsion.

### Table 10

		Likelihood Ratio Tests		
	-2 Log Likelihood of			
Effect	Reduced Model	Chi-Square	df	Sig.
Intercept	49.204	.000	0	
Gender	65.888	16.684	2	.000**
Race	51.504	2.301	4	.681

\*\* Significant at the .01 level.

Furthermore, as it relates to the multinomial logistic regression coefficients, the Wald Statistics was used to measure the contribution of gender and race to the exclusionary discipline types. The regression coefficient for the female category of gender in response category ISS of Discipline was significant (B = 2.69,  $\chi 2$  = 6.93, p = .008, suggesting that observing the female category of gender will increase the likelihood of observing the ISS category of discipline comparative to the DAEP category by 1380.12% to the male category of gender. Moreover, African American regression coefficient for ISS was (B=.266,  $\chi 2 = .371$ , P<.05), indicating that observing the African American category of race did not have a significant effect on the likelihood of discipline. Furthermore, the regression coefficient for Whites in ISS was (B=.463,  $\chi^2$ =.822, p = .365). Finally, the regression coefficient for the female category of gender in response category OSS of Discipline was significant, (B = 2.72,  $\chi 2$  = 6.95, p = .008), suggesting that observing the female category of gender will increase the likelihood of observing the OSS category of discipline relative to the DAEP category by 1420.23% compared to the male category of gender.

The regression coefficient for the African American category of race in response category OSS of discipline was not significant, (B = -0.48,  $\chi 2$  = 0.11, p =..918),

suggesting that observing the African American category of race did not have a significant impact on the likelihood of observing the OSS category of discipline relative to the DAEP category. See Table 11 for results. The overall R-squared for model 1 was .039. The results indicate that gender explains approximately 4% of the variance in the dependent variable. In model 2 the overall R-squared was .041. The results indicate that gender explains approximately 4% of the variance that

# Table 11

								95% Co	onfidence
								Inter	val for
								Ex	p(B)
			Std.					Lower	Upper
Discipl	ine <sup>a</sup>	В	Error	Wald	df	Sig.	Exp(B)	Bound	Bound
	Intercept	2.011	.354	32.237	1	.000			
	Female	2.695	1.023	6.932	1	.008	14.801	1.991	110.018
	Male	$0^{\mathrm{b}}$	•	•	0	•	•		
ISS	African	.266	.437	.371	1	.542	1.305	.554	3.072
	American								
	White	.463	.510	.822	1	.365	1.588	.584	4.317
						•		•	
	Intercept	1.261	.375	11.323	1	.001			
	Female	2.721	1.032	6.950	1	.008	15.202	2.010	114.969
	Male	$0^{\mathrm{b}}$	•	•	0	•	•		
OSS	African	048	.463	.011	1	.918	.954	.385	2.362
	American								
	White	.174	.538	.105	1	.746	1.190	.415	3.418

Parameter Estimates for the Multinomial Logistic Regression Model

Note:

a. Nagelkerke R- Squared- Model 1 (ISS) is .039 and Model 2 (OSS) is .041

b. The reference category is: DAEP

c. This parameter is set to zero because it is redundant

## Assumptions

Based on research question three for this study required logistic regression. Hence, for this study to be valid assuming one is the dependent variable it should be measured at the nominal level. For assumption two, there are one or more independent variables that are continuous, ordinal, or nominal. Additionally, for assumption two it should be independent of observations and the dependent variable should have mutually exclusive and exhaustive categories. With assumption four there is no multicollinearity among the independent variables, and in assumption five the logistic regression assumes linearity of independent variables and concludes with assumption six noting there should be no outliers, high leverage values, or highly influential points.

For this study all assumptions were met with the dependent variable having three categories: 1) DAEP, 2) In-School Suspension, and 3) Out-of-School Suspension. These were measured at the nominal level. The independent variables of Gender and Race are nominal.

## **Summary of Hypotheses Tested**

Three statistical (null) hypotheses were tested in this empirical study. Of the three hypotheses, two of them were found to be statistically significant. All three hypotheses were tested for the relationship between the variables gender, race, and exclusionary discipline types of District Alternative Placements (DAEP), In-School Suspension (ISS), and Out-of-School Suspension (OSS).

Hypothesis one revealed that there was no significant relationship between student race and occurrences of exclusionary discipline practices of the school district in the study. The variable race was composed of African American and White students, while, exclusionary disciplines were divided into three categories: DAEP, ISS, and OSS.

Furthermore, hypothesis two indicated that there was a significant relationship between gender and occurrences of exclusionary discipline practices of the school district in the study. The variable gender included female and male students.

Finally, hypothesis three was divided into two parts, a and b. Part a controlling for gender revealed that there was no significant relationship between race and occurrences of exclusionary discipline practices of the school district in the study. However, in hypothesis three, part b controlling for race, there was a significant relationship between gender and occurrences of exclusionary discipline practices of the school district in the study. The results and their relationship to the research hypotheses are presented below (See Table11).

# Table 12

Hypotheses	Value	df	Р	Conclusion
$Ho_{1:}$ There is no significant relationship between student race and the exclusionary discipline practices of the school district in the study.	3.509	4	.477	Non-Significant
Ho <sub>2:</sub> There is no significant relationship between student gender and the exclusionary discipline practices of the school district in the study.	12.615	2	.002**	Significant

# Summary of All Hypotheses Tested

# Table 12 continued

Ho <sub>3a:</sub> Controlling for gender, race does not affect on the exclusionary discipline practices of the school district in the study.	2.301	4	.681	Non-Significant
$Ho_{3b}$ : Controlling for race, gender does not affect the exclusionary discipline practices of the school district in the study.	16.684	2	.000**	Significant

\*\*Significant at the .01 level

# **CHAPTER 5**

# DISCUSSION

# Summary

The primary purpose of this study was to determine if exclusionary discipline practices and policies in the educational and criminal justice system work in a manner that adversely impacts Black students (race or gender), and to what magnitude in a Texas public school district. This chapter summarizes and discusses the overall findings of the analysis concerning a student's race, gender, and exclusionary discipline practices in a Texas public K-12 school district.

A quantitative, ex post facto research design was used in the present investigation. Archival data was retrieved for six hundred seventy-three students from a local Pre-K-12<sup>th</sup> public Texas school district. The data encompassed exclusionary discipline types (ISS, OSS, and DAEP) that were categorized by race and gender during the 2016-2017 school year. Finally, the data were analyzed using Multinomial Logistic Regression to assess whether race and gender had a significant effect on the likelihood of observing each response classification of discipline to DAEP. Multinomial logistic regression was also performed to model the relationship between the predictors and membership in the three groups (race, gender, and exclusionary discipline), with the disciplinary infractions being the dependent variable. The study was guided by the following research questions:

- RQ 1: Is there a significant relationship between student race (Black and White) and exclusionary disciplinary practices within a Texas public school district?
- RQ 2: Is there a significant relationship between student gender (Male or Female) and exclusionary disciplinary practices within a Texas public school district?
- RQ 3: Is there a significant relationship between student race (Black and White), student gender (Male or Female), and exclusionary discipline practices within a Texas public school district?

In addition, the following null hypotheses were formulated and tested in this study:

- Ho<sub>1</sub>: There is no statistically significant relationship between student race (Black or White) and exclusionary disciplinary practices.
- Ho<sub>2</sub>: There is no statistically significant relationship between student gender and the exclusionary discipline practices of the school district in the study.
- Ho<sub>3</sub>a: Controlling for gender, race does not affect the exclusionary discipline practices of the school district in the study.
- Ho<sub>3</sub>b: Controlling for race, gender does not affect the exclusionary discipline practices of the school district in the study.

# **Findings**

The following findings were generated from the results of this investigation:

- No significant relationship between student race and exclusionary disciplinary practices was found. While Black students in total receive more exclusionary disciplinary actions, the number is not disproportionate to what would be expected from the ratio of Black students to White students in the analysis.
- 2. A relationship was found to exist between student gender and exclusionary discipline practices.
- Race does not affect the exclusionary discipline practices when gender is controlled for the school district in the study.
- 4. Significant unique contributions were made by gender, but not for race. For gender, the likelihood of a female being in the ISS group rather than the exclusionary discipline practices increased, and the likelihood of females being in the OSS group rather than the exclusionary discipline practices group also increased.

#### **Discussion**

One of the most significant findings of the present study was that there was no relationship between race and exclusionary disciplinary practices of the school district in the study. This finding was inconsistent with those of Nowick (2018) and Cardichon and Hammond (2019), who documented in the past three decades that African American students, especially African American males are, overrepresented when it involves exclusionary discipline, particularly in suspension and expulsion. Similarly, the U.S Department of Education Office for Civil Rights (2018) discovered during the 2013–2014 school year, that Black students from pre-K to 12th were 3.8 times more likely to have one or more out-of-school suspensions than White students. According to a study by Skiba et al. (2008), exclusionary discipline practices are disproportionately applied to students of color. In addition, Losen (2015) found that students of color are suspended from school for minor offenses that do not pose a serious threat to school safety. African American students receive suspensions for less serious violations in comparison to their White peers. Equally, Steinberg and Lacoe (2017) asserted that Black students are more likely than White students to receive suspension or expulsion for minor offenses.

Literature involving discipline disparities has existed for many decades from Bland and Harwin (2017) reporting data from the Department of Education's Civil Rights Database regarding significant increases in racial disparities for Black student arrests and referrals to the police. Bouchein (2015) to TABSE (2019) noted the negative implications of zero-tolerance policies for public-school students, especially African American and special education students. Moreover, Rausch and Skiba (2004) discovered that schools with higher rates of suspension reported a drop in scores related to academic achievement when controlling for poverty and race. Steinberg and Lacoe (2017) claimed racial disparities in suspension begin as early as preschool with Black students. The aforementioned studies reject the results of the present study. A plausible explanation of this finding is that states other than Texas have adopted less punitive approaches as well as reduced the use of exclusionary practices to ensure a more inclusive learning environment for all students. Another finding of the present study was a significant relationship between gender and exclusionary discipline practices. Consistent with the findings of Raffaele Mendez et al. (2002), Black girls have a higher percentage of suspensions than Hispanic or White girls. An examination of exclusionary discipline sanctions, such as in-school and out-of-school suspensions, was reported in a study by Bale et al. (2011). In all discipline sanctions, Black females were disproportionately affected. Blake and colleagues found that Black girls were twice as likely to receive an in-school suspension and out-of-school suspension as their same-gender peers. According to Skiba et al. (2002), studies on Black female discipline experiences have mostly examined discipline sanctions of Black boys compared to Black girls, with Black girls rarely being mentioned outside of descriptive statistics.

Intriguingly, the narrative involving gender disproportions was drastically dissimilar from the narrative concerning racial disparities for this analysis. The body of literature suggests that there are exclusionary discipline disparities across gender lines that involve boys experiencing disciplinary action far more frequently than girls (Gregory, 1997; McFadden et al., 1992; Shaw and Braden, 1990). The National Center for Education Statistics (2016) reported that female African American students enrolled in grades 6 through 12 had been suspended 29% more times than White students enrolled in grades 6 through 12. The results of the present study extend what is already known regarding gender and exclusionary disciplinary practices. This research study adds to existing research, which acknowledges that gender disparities exist but does not propose solutions to decrease African American exposure to the juvenile justice system.

An explanation for the gender finding may be that the literature appears to report Black females' discipline sanctions as a way to draw attention to the disparity between Black males' and Black females' discipline experiences. School discipline literature has given limited attention to Black females compared with Black males.

Moreover, the most surprising finding of the present study was the lack of evidence suggests race is a more significant predictor of disciplinary outcomes than gender (Skiba, Chung, et al., 2014). Skiba, Chung, et al. (2014) found that race was a significant predictor of OSS and expulsion regardless of gender. Contrary to the findings of this study, Steinberg and Lacoe (2017) also asserted that Black and Hispanic students are more likely than White students to receive suspension or expulsion for minor offenses. Similarly, the U.S Department of Education Office for Civil Rights (2018) discovered during the 2013–2014 school year that Black students from pre- k to 12th were 3.8 times more likely to have one or more out-of-school suspensions as White students were.

Likewise, Rocque (2010) found evidence of a Black-White disciplinary gap in disciplinary referrals. Gregory and Weinstein (2008) and Skiba et al. (2014) reported racial disparities concerning ISS and OSS suspensions.

As it relates to research question 3, it was discovered that disparities existed with females receiving a higher rate of ISS and OSS than males, and the likelihood of a female being assigned to ISS and OSS was greater for this particular school district. Although race was not significantly associated with the exclusionary disciplinary practices of the school district when controlling for gender, gender was significantly associated with the exclusionary disciplinary practices of the school district when controlling for race. To forge one's thinking and extend to previous research as it relates to the literature review for this analysis, it would be conducive to explore why for this particular school district females receive the lesser consequences of exclusionary discipline in comparison to their male counterparts, and why females received a higher rate of suspension than their male counterparts.

An explanation of why this particular school district's findings did not align with other studies and researchers could be: what does this school district consider an offense that warrants an office referral, most importantly the question is, what is the reported data not informing us as researchers or what circumstances arise on campus that does not automatically warrant a discipline referral, which could inform us about approaches which could create similar outcomes for other school districts that mirror this school districts.

# Limitations

In any research design, there are risks to internal and external validity. Internal validity refers to the degree to which changes in the dependent variable are directly connected to the independent variable. Risks to external validity consist of the ability to create generalizations from the outcomes of the analysis. The limited population of the analysis and sample size could be instances of a risk to the external validity. The sampling structure consisted of students in Grades 3rd to 12th grade within one Texas public school district for the 2016-2017 academic school year. Therefore, the findings and outcomes may not be relevant to the other 1029 Texas public school district, which results may not be generalizable to other districts or states.

More than likely, some students in the sample are represented more than once in the dataset. For example, some students had repeat offenses that are not accounted for in the analysis. Since the number of repeated offenses was not controlled, the impact that this repetition might have on disciplinary actions taken is unclear. For example, if a student receives five disciplinary referrals for the same offense, he or she may receive a more severe consequence than a student whose first offense is that offense. Additionally, because I did not control for the offense that led to the disciplinary action, I am not confident in saying that the severity of offense with the same disciplinary actions is equivalent.

Last, school district administrators are responsible for assigning discipline and also responsible for issuing exclusionary discipline (ISS, OSS, DAEP) consequences to students are guided by outside factors, including state guidelines, district expectations, and personal bias. The race of the teachers and administrators was not noted, which limited analysis of its influence on the use of exclusionary discipline within this Texas public school district. This study was intended for state legislators, school districts, and campuses in addition to adding to the literature as it relates to the adverse impact of exclusionary discipline practices.

# Implications

The purpose of this study is not to categorize exclusionary discipline practices as effective or ineffective as they relate to student referrals. The findings raise awareness about exclusionary discipline practices and their relationship to a student's race and gender. The findings in chapter 4 have implications for exclusionary discipline practices and policies within this and perhaps other Texas school districts. Further, my analysis of the data is as follows:

- 1. Findings from this study could increase awareness among school leaders of the harmful effects any level of exclusionary discipline may have on the academic achievement of specific demographic and ethnic groups. From this current study campus leaders should consider addressing the impact that exclusionary discipline practices, such as ISS and OSS have on females and the impact DAEP has on males (See table 9 and 13).
- 2. Limit administrators' discretion in exclusionary discipline (OSS and ISS) actions-consequences for this school district. As noted above, the analysis discovered that female students received exclusionary discipline practice of ISS or OSS at a higher rate than their male counterparts, and male students received exclusionary discipline practice of DAEP at an alarming rate in comparison to their female counterparts for this particular school district. Exclusionary Discipline practices should be broadened to include a discontinuance or other alternatives concerning the application of ISS and OSS for a violation involving discretionary referrals. An alternative could include eradicating the use of ISS and OSS for lower-level student code of conduct infractions such as disrespect towards adults or violating school rules (causing a disruption during instructional time or dress code violation). An option to ISS and OSS could include non-exclusionary prevention measures such as referral to a behavioral specialist and or a counselor. For certain circumstances, it may warrant before or after-school detention.

3. To minimize the effects of exclusionary discipline practices on this population researchers adamantly favor implementing restorative discipline practices. The analysis is consistent with Pavelka's (2013) application for further research regarding the use of restorative justice as a principle-based model. The restorative discipline model seeks to determine the impact of the behavior and establish a mutual, prescriptive agreement for repairing the harm caused by the wrongdoing (p. 15). Restorative justice is based on three core principles [which include] Repair harm. Restorative justice requires that victims and communities are healed of the harm which resulted from the wrongful occurrence. Wrongdoers are held accountable for their actions and encouraged to make positive changes in their behavior. Reduce risk. Community safety requires practices that reduce risk and promote the community's capacity to manage behavior. Citizens feel safe and can live in peace when wrongful behavior is prevented and controlled.

Empower community. Schools, along with the external community, must take an active role in and responsibility for the restorative response by collectively addressing the impact of the wrongdoing and the reparation. Students are empowered as active participants in the resolution process (Pavelka, 2013, p. 15).

## **Suggestions for Future Practice**

The results of this analysis support existing research related to exclusionary discipline practices and gender as it relates to males' overrepresentation in DAEP and girls' overrepresentation in ISS and OSS. Additionally, the analysis contributed to the

body of knowledge by seeking to understand the variance in the exclusionary discipline practices within a small Texas school district and to determine if there is an association between exclusionary discipline and students' race and gender at a local school district. Moreover, the present analysis examined the percentage of exclusionary discipline assignments as it relates to African American students and the combination of another race and gender to determine if there was a statistically significant variance. This analysis had the potential to serve as a beginning stage in addressing ways to decrease the use of exclusionary discipline practices of DAEP when it relates to male students, and reduce the application of ISS and OSS for females at this particular school district.

In addition, as many are aware, schools are responsible for ensuring that many things transpire throughout the school day. Creating a safe environment that does not encompass disruptive behaviors and one that does not impede the learning process is paramount to the success of all students. Educators also are held responsible for creating a supportive campus atmosphere, all while cultivating a positive learning environment for all students. The restorative discipline approach allows students to make a mistake and learn from it, remain in the learning environment, and actively engage in the instructional content presented in the classroom. I recommend that this practice be incorporated into this school district. Additionally, previous research has noted the favorable outcomes of the use of the restorative discipline model as it is rooted in building positive relationships with students and others within the school community (Colombi & David, 2015).

Future research in this field of study should also consider the effects of being excluded from the educational environment as well as the impact of the amount of time students are away from the educational environment. For example, do two days of OSS or ISS have more or less adverse impact as it relates to a student's educational outcome? In fact, some states tend to establish suspension limits capriciously; in Texas, OSS may not exceed three instructional days (Texas Education Code Sec. 37.005(b)), whereas in California the limit is 20 days over a school year (California Education Code Sec. 48903), and in Florida suspension may not exceed 10 instructional days (Florida Education Code Sec. 1003.01 5(a)). Moreover, the research needs to be completed concerning the length of exclusion from the educational environment to more intentionally establish state- and district-level exclusion guidelines and boundaries.

Furthermore, future research should closely examine the relationship between exclusionary discipline and the use of this practice application based on categories of race and gender in both small and large school districts. Research advises that exclusionary disciplinary practices have adversely impacted students in ways that essentially cause the student to enter into the juvenile and or criminal justice system. In addition, another recommendation includes conducting a study looking into discipline practices at the campus level and interviewing campus administrators regarding their exclusionary discipline practices. As such, a mixed method analysis may provide further understanding as it relates to why male students are sent to DAEPs at a higher rate than their female counterparts.

Future recommendations should also include and focus on the educational experiences of female students who have experienced exclusionary discipline at a rate alarmingly higher than their male counterparts. It was noted that nationally Black girls often experience exclusionary discipline outcomes more than their male counterparts, which is a trend that is aligned within the criminal justice system (Chesney-Lind, 2010; Tate et al., 2014).

As it relates to recommended analysis, this should include longitudinal data examination of students' exact discipline data. Student exact data would permit analysis of the usefulness of suspensions through the occurrence of recurring suspendable infractions. Longitudinal data that follow students through their years in the school district would permit analysis of the future impact of suspension and possibly assist in identifying predictive influences regarding younger students. Secondly, an analysis should be conducted to determine if a student's grade level impacts exclusionary discipline practices. In addition to these specific research intentions, it can be safely noted that there remains much to be learned about disciplining within this Texas public school district and, across the U.S.

Last, although not analyzed in this study, a meaningful topic for examination is the suspension percentage of students in the categories of race and gender as it relates to their respective grade levels. There are extremely too many schools relying on exclusionary discipline that are a result of ZTPSs that have essentially caused longlasting consequences. Further studies that would more closely examine exclusionary practices in conjunction with the gender of students would greatly benefit current and future educators. Although there was no report of expulsions in this analysis, it is suggested that the analysis be duplicated in a school district that is larger than the district analyzed in this study and one that has a large number of expulsions. With the suggested approach, this is the only way the scientific community can decide the differences in suspensions related to a student's race and the combination of gender to determine if this is consistent with expulsions.

#### **Recommendation for Policy and Practice**

Implementation of Data Driven Decisions. Districts should place limits on the use of exclusionary discipline practices as it relates to a student's first discretionary infraction, especially given that the referral significantly increases the student's risk of future offenses. Other non-exclusionary best practices should be implemented as they relate to disciplinary practices for first referrals. Another policy recommendation is to eradicate and or revisit the use of zero-tolerance policy and exclusionary discipline practices in Texas public schools. The overall goal is for state and local officials to eradicate or revisit the use of overly harsh discipline policies such as OSS, ISS, and DAEP for minor or repeated behaviors (e.g., being disrespectful, chewing gum, and skipping class.) State and local officials should strive to offer restorative discipline approaches that restore a student's behavior as opposed to utilizing practices that will increase a scholar's chances of coming in contact with the juvenile or adult legal system.

Implement an alternative restorative approach to decrease student misconduct. It does not follow from these findings; neither was it the intent of the present analysis to classify exclusionary discipline as an "advantage" or "disadvantage" practice as it relates to students' particular violations. Instead, the purpose of this study was to enlighten all of the relationships between exclusionary discipline practices regarding their application based on the categories of race and gender within a Texas school district. As it relates to placing limits on the exclusionary discipline practices of OSS, ISS, and DAEP removals, campuses and school districts should consider implementing campus- and district-wide approaches to school discipline that improve the overall educational outcomes for students that have been identified as at-risk. Moreover, the purpose of this study was to shed light that would guide educators and administrators to be more thoughtful and suitable in their referral and processing of discipline infractions.

**Practice Recommendation.** Colleges, teacher alternative certification programs, and public-charter schools should create a partnership to construct a required curriculum. With that, the curriculum should be centered around cultural awareness and competency modules for their respective training courses for teachers and school leaders using the critical pedagogy framework to enhance the learning experiences of all learners, with a special focus on African American scholars. Addressing cultural biases within the educational setting will essentially attempt to resolve the overreliance on exclusionary discipline practices. In addition to addressing the overreliance on exclusionary discipline, this could also serve in building positive relationships with students that result in increasing their time in the class as opposed to the office for a referral.

## Conclusion

Existing research on disproportionality in the assignments of exclusionary discipline has a variety of emphases. Some are concerned with corroborating the existence of disproportions across a multitude of settings. Frequently, this analysis exemplifies large-scale quantitative studies that make assumptions from large exclusionary discipline data. Numerous school districts frequently compile this kind of data and many states including Texas mandate school districts to report this data. There is an overabundance of research that notes the adverse consequences of exclusionary

discipline practices from a state standpoint but not necessarily from a local school district stance.

This study intended to identify the disparities that occur in a single school district, and to investigate if a student's race or gender has any impact on exclusionary discipline practices. This analysis' research questions exemplify its focus 1. Does race have an impact when it relates to exclusionary discipline in one Texas public school district? 2. Does race have an impact when it relates to exclusionary discipline in one Texas public school district? 3. Does a student's race or gender have an impact on exclusionary discipline practice in a Texas public school district? The analyses conducted assumed the outcomes in the district would be the representation of national trends. The analysis research questions rest firmly on the intended purpose.

Moreover, the outcome of the study contributes to the body of knowledge by recognizing the variables of race and gender in one Texas public district. Future studies should look deeply at males' overrepresentation in DAEP and females' overrepresentation in the discipline categories of ISS and OSS, especially for this school district. Educators appear to believe that sustaining exclusionary discipline practices in school is critical to ensuring a safe learning environment. However, the disaster of punitive school discipline policies such as ZTPs has negatively impacted all to a significant degree across the state and nation.

With other disciplinary best practices, campuses and school districts can leverage the necessity of a safe, cultivating, and fruitful learning environment with the main focus of educating students. All students deserve a chance to obtain a high-quality education, so it is essential that educators and policymakers guarantee that academic achievement is the main objective for all students. The school-to-prison pipeline remains a significant issue in the United States. Existing research has established a strong framework upon which future studies for this social phenomenon can be conducted. Also, findings from the present study suggest that school districts should adopt practices to decrease the use of exclusionary discipline for females as well as conduct further note future research to track the use of non-exclusionary discipline and the particular infractions to determine if there are any disparities. Educational leaders and local and state leaders would benefit from implementing a universal method of teacher preparation, which should include culturally responsive instruction as well as aggressively monitoring and revising the zerotolerance policies.

One cannot overlook what is apparent, and that is young scholars are suspended and expelled at alarming rates, which essentially does not always support their academic success. With unconventional disciplinary strategies, schools can balance the need for a safe, caring, and positive learning environment with the primary goal of educating youth. All young people deserve an opportunity to receive a high-quality education. It is crucial that educators and policymakers assure that academic success is the main goal for all students.

#### REFERENCES

- 2013-2014 Civil Rights Data Collection (2016). Retrieved from https://www2.ed.gov/about/offices/list/ocr/docs/2013-14-first-look.pdf
- Aizer, A., & Doyle Jr., J. (2015). Juvenile incarceration, human capital and future crime: Evident from randomly- assigned judges. *The Quarterly Journal of Economics*, 130(2), 759-803.
- Alexander, R. J. (2010). Children, their World, their Education: final report and recommendations of the Cambridge Primary Review, London, Routledge.
- American Academy of Pediatrics. (2013). Out-of-School Suspension and Expulsion. 131(3). Retrieved from: https://doi.org/10.1542/peds.2012-3932
- American Civil Liberties Union (2019). School-to-Prison Pipeline Smart Justice: What's at stake. https://www.aclu.org/issues/smart-justice
- American Psychological Association (2008). Are Zero Tolerance Policies Effective in the Schools? 63 (9),852-862. doiI:10.1037/0003-066X.63.9.852
- Annamma, S. A., Anyon, Y., Joseph, N. M., Farrar, J., Greer, E., Downing, B., & Simmons, J. (2019). Black girls and school discipline: The complexities of being overrepresented and understudied. *Urban Education*, 54(2), 211-242.
- Annamma, S. A., & Handy, T. (2021). Sharpening Justice Through DisCrit: A Contrapuntal Analysis of Education. Educational Researcher, 50(1), 41–50. https://doi.org/10.3102/0013189X20953838
- Anti-Defamation League (2014). Brown v. Board of Education 60 years later 1954-2014. Retrieved from https://www.adl.org/media/4783/download

- Austin, J. (2018). Texas faces questions about suspension, counselors, SROs, and schoolto-prison pipeline. The Huntsville Item. Retrieved from https://www.itemonline.com/news/texas-faces-questions-about-suspensionscounselors-sros-and-school-to/article\_14a14aa4-9864-5ecb-b9d8beaa2509339d.html
- Balfanz, R., Byrnes, V., & Fox, J. (2013). Sent home and put off-track: The antecedents, disproportionalities, and consequences of being suspended in the ninth grade.
  Paper presented at the conference Closing the School Discipline Gap: Research to Practice, Washington, DC.
- Balfanz, R., Spiridakis, K., Neild, R. C., & Legters, N. (2003). High-poverty secondary schools and the juvenile justice system: How neither helps the other and how that could change. *New Directions for Youth Development, 2003*(99), 71–89. https://doi.org/10.1002/yd.55

Balingit, M. (2018). Racial disparities in school discipline are growing, federal data show. The Washington Post. Retrieved from https://www.washingtonpost.com/local/education/racial-disparities-in-school-discipline-are-growing-federal-data-shows/2018/04/24/67b5d2b8-47e4-11e8-827e-190efaf1f1ee\_story.html

Basler, J. (2012). *The moral development and perceptions of academic and clinical integrity of nursing students: An ex post facto study*. (Doctoral dissertation).
Retrieved from ProQuest Dissertations and Theses. (Accession Order No. 3540440).

- Beck, A. N., & Muschkin, C. G. (2012). The enduring impact of race: Understanding disparities in student disciplinary infractions and achievement. *Sociological Perspectives*, 55(4), 637–662. https://doi.org/10.1525/sop.2012.55.4.637
- Bell, D. A. (1976). Serving two masters: Integration ideals and client interests in school desegregation litigation. *Yale Law Journal*, 85, 470–516.
- Besse, R., & Capatosto, K. (2018, May). Ending racial inequity in out-of-school suspensions mapping the policy landscape and equity impact. *Kirwan Institute for the Study of Race and Ethnicity*, 1-14. Retrieved from http://kirwaninstitute.osu.edu/implicit-bias-training/resources/OSS-racialinequity-02.pdf

Bever, L. (2016). The root of discipline disparities. public school forum. https://www.ncforum.org/the-root-of-discipline-disparities/
Black Men's College Enrollment. The Review of Black Political Economy, 00346446211036763. https://doi.org/10.1177/00346446211036763

- Blackburn, D. (2016). The problem with zero-tolerance policies. *Texas Monthly*. Retrieved from https://www.texasmonthly.com/the-daily-post/zero-tolerance-policies-asthma-attack/
- Blake, J. J., Butler, B. R., Lewis, C. W., & Darensbourg, A. (2011). Unmasking the inequitable discipline experiences of urban Black girls: Implications for urban stakeholders. *Urban Review*, 43(1), 90-106.
- Bland, E., & Harwin, A. (2017). Black students more likely to be arrested at school.*Education Week, Week 22* (21). Retrieved December 23, 2019, from

https://www.edweek.org/ew/articles/2017/01/25/black-students-more-likely-tobe-arrested.html

- Boghani, P. (2015). How poverty can follow children into adulthood. Front Line Journalism Fund. Retrieved from https://www.pbs.org/wgbh/frontline/article/how-poverty-can-follow-childreninto-adulthood.
- Booker, K., & Mitchell, A. (2011). Patterns in recidivism and discretionary placement in disciplinary alternative education: The impact of gender, ethnicity, age, and special education status. *Education & Treatment of Children, 34*(2), 193–208. https://doi.org/10.1353/etc.2011.0016
- Bornstein, B., & Miller, M. (2019). *Advances in psychology and law (4th ed.)*. Reno, NV: Springer.
- Bouchein, M. (2015). School-to-prison pipeline: A comparison in discipline policy between Maryland and Texas public schools. *Maryland Equity Project Advancing Educational Opportunities*, 1-20.
- Bristol, T. J., & Mentor, M. (2018). Policing and teaching: The positioning of Black male teachers as agents in the universal carceral apparatus. Urban Review: Issues and Ideas in Public Education, 50(2), 218-234.

Britton, T. A. (2021). Does locked up mean locked out: The effects of the 1986 anti-drug abuse act on college enrollment. *Journal of Economics, Race, and Policy*, 5, 54–71. https://doi.org/10.1007/s41996-020-00073-y

- Britton, T.A. (2021). Educational opportunity and the carceral system: Sentencing policies and black men's college enrollment. *The Review of Black Political Economy.* https://doi.org/10.1177/00346446211036763
- Brophy, J. (1988). Research linking teacher behavior to student achievement: Potential implications for instruction of chapter 1 students. *Educational Psychologist*, 23(3), 235. https://doi.org/10.1207/s15326985ep2303\_3

Brown v. Board of Education of Topeka, 347 U.S. 483 (1954).

Brown, C. A., & Di Tillio, C. (2013). Discipline disproportionality among Hispanic and American Indian students: Expanding the discourse in U.S. research. *Journal of Education and Learning*, 2(4), 47–59. https://doi.org/10.5539/jel.v2n4p47

Bryan, J., Day-Vines, N., Griffin, D., & Moore-Thomas, C. (2012). The disproportionality dilemma: Patterns of teacher referrals to school counselors for disruptive behavior. *Journal of Counseling & Development*, 90(2), 177-190. DOI:10.1111/j.1556-6676.2012.00023.x

- Busby, D. R., Lambert, S. F., & Lalongo, N. S. (2013). Psychological symptoms linking Exposures to community violence and academic functioning in African American adolescents, *Journal of Youth Adolescence*, 42(2), 250-262. Retrieved from http://dx.doi.org/10964-012-9895-z
- Cardichon, J., & Darling-Hammond, L. (2019). *Protecting students' civil rights: The federal role in school discipline*. Palo Alto, CA: Learning Policy Institute.
- Carmichael, D., Whitten, G., & Voloudakis, M. (2005). Study of minority
   overrepresentation in the Texas Juvenile Justice System, final report. The Public
   Policy Research Institute. Texas A&M University.

- Carson, E. (2020). Prisoners in 2018. U.S. Department of Justice Office of Justice Programs.
- Carter, P. L. (2018). Education's limitations and its radical possibilities. *Contexts*, *17*(2), 22–27.

Cerrone, K. (1999). The gun-free schools Act of 1994: Zero Tolerance takes aim at procedural due process, 20 Pace L. Rev. 131. https://digitalcommons.pace.edu/plr/vol20/iss1/7

- Chesney-Lind, M. (2010). Jailing "bad" girls: Girls' violence and trends in female incarceration. In M. Chesney-Lind & N. Jones (Eds.), *Fighting for girls*. New York: State University of New York Press, 57-79
- Cohen, R. (2016, Fall). Rethinking school discipline. *The American Prospect magazine*. Retrieved from https://prospect.org/education/rethinking-school-discipline/
- Colombi, G., & Osher, D. (2015). Advancing school discipline reform Education Leaders Report. Arlington, VA National Association of State Boards of Education.
- Costenbader V., & Markso,n S. (1998). School suspension: A study with secondary school students. *Journal of School Psychology*. *36*, 59–82.
- Cradle to Prison Pipeline Factsheet Michigan. (2009). Children's Defense Fund. Retrieved February 26, 2014, from http://www.childrensdefense.org/childresearch-data-publications/data/state-data-repository/cradle-toprisonpipeline/cradle-prison-pipeline-michigan-2009-fact-sheet.pdf.
- Craven, M. (2017). Banning classroom removals for young children Policy guide. Texas Appleseed. Retrieved from http://www.texasappleseed.org/sites/default/file/Yg-Stud-SuspensionBanPolicyGuide.pdf

- Craven, M. (2017). Suspended childhood an analysis of exclusionary discipline of Texas' pre-K and elementary school students. Texas Appleseed. Retrieved from http://stories.texasappleseed.org/suspended-childhood-updated
- CRDC. (2016). A first look. Washington, D.C.: United States Department of Education.
- Crenshaw, K. W., Ocen, P., & Nanda, J. (2016). Black girls matter: Pushed out, over policed and under protected. African American Policy Forum, Center for Intersectionality and Social Policy Studies.
- Crichlow, W. (2015). Critical race theory: A strategy for framing discussions around social justice and democratic education. Higher Education in Transformation Conference, Dublin, Ireland, 2015,.187-201
- Cuellar, A.E., & Markowitz, S. (2015). School suspension and the school-to-prison pipeline. *International Review of Law and Economics*, *43*, 98-106.
- Curran, F. C. (2016). Estimating the effect of state zero tolerance laws on exclusionary discipline, racial discipline gaps, and student behavior. *Educational Evaluation and Policy Analysis, 38*, 647–668. doi:10.3102/0162373716652728
- Curtis, A. (2014). Tracing the school to prison pipeline from zero tolerance policies to juvenile justice dispositions. *Georgetown University Law Journal, 102*(4), 1251-1277.
- de Brey, C., Musu, L., McFarland, J., Wilkinson-Flicker, S., Diliberti, M., Zhang, A., Branstetter, C., & Wang, X. (2019). Status and trends in the education of racial and ethnic groups 2018 (NCES 2019-038). U.S. Department of Education. Washington, DC: National Center for Education Statistics. Retrieved from https://nces.ed.gov/ pubsearch/.

DeCuir, J. T., & Dixson, A. D. (2004). So when it comes out, they aren't that surprised that it is there: Using critical race theory as a tool of analysis of race and racism in education. *Educational Researcher*, 33(5), 26–31.

https://doi.org/10.3102/0013189X033005026

- Delgado, R. (1989). Storytelling for oppositionists and others: A plea for narrative. *Michigan Law Review*, 87(8), 2411–2441.
- Delgado, D., & Stefancic, J. (2001). Critical race theory: An introduction. New York: NYU Press.
- Delgado, R. (2009). Affirmative action as a majoritarian device: Or do you really want to be a role model? In Foundations of Critical Race Theory in Education (pp. 109-116). New York: Routledge. (Original work published 2009).
- Diliberti, M., Jackson, M., Correa, S., Padgett, Z., National Center for Education
  Statistics (ED), & American Institutes for Research (AIR). (2019). Crime,
  violence, discipline, and safety in U.S. public schools: Findings from the School
  Survey on Crime and Safety: 2017-18. First Look. NCES 2019-061. National
  Center for Education Statistics.
- Dixson, A.D., & Rousseau, C. (2005). And we are still not saved: Critical race theory in education ten years later. *Race Ethnicity and Education*, 8(1), 7-27, DOI:10.1080/1361332052000340971
- Eden, M. (2018). Investigation: In NYC school where a teenager was killed, students & educators say lax discipline led to bullying, chaos, and death. Retrieved from https://www.the74million.org/article/investigation-in-new-york-city-school-

where-a-teenager-was-killed-students-educators-say-lax-discipline-led-tobullying-chaos-and-death/.

- Ekstrom, R. B., Goertz, M. E., Pollack, J. M., Rock, D. A. (1986). Who drops out of high school and why? Findings from a national study. *Teachers College Record*, 87(3), 356-373.
- Epstein, R. (2014). Civil Rights Enforcement Gone Haywire. EducationNext,14 (4), 1-6. Retrieved from https://www.educationnext.org/civil-rights-enforcement-gone-haywire/
- Equal Justice Initiative (2014). The Superpredator Myth, 20 Years Later [online] Alabama: Equal Justice Initiative. Retrieved from https://eji.org/news/superpredator-myth-20-years-later/
- Fabelo, T., Thompson, M. D., Plotkin, M., Carmichael, D., Marchbanks, M. P., & Booth,
  E. A, (2011). Breaking schools' rules: A statewide study of how school discipline
  relates to students' success and juvenile justice involvement. New York, NY:
  Council of State Governments Justice Center. Retrieved from
  http://justicecenter.csg.org/resource
- Flannery, M. (2015). The school-to-prison pipeline: Time to shut it down. *National Education Association*. Retrieved from: https://www.nea.org/advocating-forchange/new-from-nea/school-prison-pipeline-time-shut-it-down
- Foiles Sifuentes, A. M. (2015). Performing the grade: Urban Latino youth, gender performance, and academic success. Race Ethnicity and Education, 18(6), 764– 784. https://doi.org/10.1080/13613324.2012.759933

- Ford, J. E. (2016). The root of discipline disparities. *Educational Leadership*, 74(3), 42-46.
- Fowler, D. (2011). School discipline feeds the "pipeline to prison." *Phi Delta Kappan*, 93(2), 14–19.
- Freire, P. (1970). *Pedagogy of the oppressed* (M. B. Ramos, Trans.) New York, NY: Continuum.
- Gagnon, D. J., Jaffee, E. M., & Kennedy, R. (2016). Exclusionary discipline highest in New Hampshire's urban schools suspension and expulsion found to disproportionately affect disadvantaged students. *The Carsey School of Public Policy at the Scholars' Repository*. 267.https://scholars.unh.edu/carsey/267
- Gay, L., Mills, G., & Airasian, P. (2016). Educational research: Competencies for analysis and applications. Upper Saddle River, NJ: Merrill/Pearson.
- Giroux, H. (2013). *America's education deficit and the war on youth*. New York, New York.
- Gillborn, D. (2006). Critical race theory and education: Racism and anti-racism in educational theory and praxis. *Discourse: Studies in the Cultural Politics of Education*, 27(1), 11–32. https://doi.org/10.1080/01596300500510229
- Gilliam, W.S., Maupin, A.N., Reyes, C.R., Accavati, M., & Shic, F. (2016). Do early educator's implicit biases regarding sex and race related to behavior expectations and recommendations of preschool expulsions and suspensions? *Yale Child Study Center, September*, 991-1013.
- Gordon, N. (2018, January). *Disproportionality in student discipline: Connecting policy to research.* Brookings.

- Gregory, A., & Weinstein, R. S. (2008). The discipline gap and African Americans:Defiance or cooperation in the high school classroom. *Journal of SchoolPsychology*, 46, 455–475.
- Gregory, A., Hafen, C. A., Ruzek, E., Mikami, A. Y., Allen, J. P., & Pianta, R. C. (2016).
  Closing the racial discipline gap in classrooms by changing teacher practice.
  School Psychology Review, 45(2), 171–191.
- Gregory, A., Skiba, R. J., & Noguera, P. A. (2010). The achievement gap and the discipline gap: Two sides of the same coin? *Educational Researcher*, 39(1), 59–68. https://doi.org/10.3102/0013189X09357621
- Gregory, A., Skiba, R. J., & Noguera, P. A. (2010). The achievement gap and the discipline gap: Two sides of the same coin? *Educational Researcher*, 39(1), 59-68.
- Gregory, A., & Weinstein, R. S. (2008). The discipline gap and African Americans:
  Defiance or cooperation in the high school classroom. *Journal of School Psychology*, 46(4), 455–475. doi:10.1016/j.jsp.2007.09.001.
- Haight, W.L., Kayama, M., & Gibson, P.A. (2016). Out-of-school suspensions of black youths: Culture, ability, disability, gender, and perspective. *Social Work*, *61*(3), 235-43. https://doi.org/10.1093/sw/sww021
- Hancock, S. D., & Warren, C. A. (2017). *White women's work: Examining the intersectionality of teaching, identity, and race.* Information Age Publishing, Inc.
- Harper, K. (2008). Are zero tolerance policies effective in the schools? *AmericanPsychological Association Zero Tolerance Task Force*, 63(9), 852-862.

- Harper, K., Ryberg, R., & Temkin, D. (2018). Schools report fewer out-of-school suspensions, but gaps by race and disability persist. Child Trends.
  https://www.childtrends.org/schools-report-fewer-out-of-school-suspensions-but-gaps-by-race-and-disability-persist.
- Harper, K. (2019). The need for a comprehensive approach to address school discipline & discipline disparities by race & disability in Pennsylvania. Child Trends.
  Received from https://www.childtrends.org/publications/the-need-forcomprehensive-approach-address-school-discipline-discipline-disparities-by-racedisability-in-pennsylvania
- Heckman, J. J., & LaFontaine, P. A. (2010). The American high school graduation rate: Trends and levels. *Review of Economics and Statistics*, 92(2), 244–262. doi.org/10.1162/rest .2010.12366
- Heitzeg, N. A. (2009). Education or incarceration: Zero tolerance policies and the school to prison pipeline. *Forum on Public Policy Online*, 2009(2), 1-21.
- Heitzeg, N. (2016). The School-to-Prison Pipeline: Education, Discipline, and Racialized Double. Santa Barbara, California: ABC-CLIO, LLC. *in Education* (pp. v-xiii) [Foreword]. New York: Routledge.
- Irby, D. J. (2014). Trouble at School: Understanding School Discipline Systems as Nets of Social Control. *Equity & Excellence in Education*, 47(4), 513–530. https://doi.org/10.1080/10665684.2014.958963
- Jemal A. (2017). Critical consciousness: A critique and critical analysis of the literature. *The Urban Review*, 49(4), 602–626. https://doi.org/10.1007/s11256-017-0411-3

- Jiminez, L., & Flores, A. (2019). 3 Ways DeVos has put students at risk by deregulating education. Center for American Progress, 1-9. Retrieved from https://www.americanprogress.org/issues/education-k-12/reports/2019/05/30/470509/3-ways-devos-put-students-risk-deregulatingeducation/
- Johnson, J., Wilson, S., & Greene, Taylor, H. (2015 unpublished paper). An Exploratory Analysis of the School to Prison Pipeline and Disproportionate Minority Contact in Texas.
- Johnson, K., & Williams, D. (2020). Zero-tolerance discipline disproportionately affects students of color, with disabilities. *The Source*. Retrieved from https://www.tpr.org/post/zero-tolerance-discipline-disproportionately-affectsstudents-color-disabilities
- Johnson, R. (2016). Zero tolerance policies likely contribute to high attrition rates of Black students and Hispanic students. Intercultural Development Research Association. Retrieved from http://www.idra.org/wp-

content/uploads/2016/11/IDRA-Discipline-Article-2016.pdf

Johnson, S. (2002, June). Diversity policy in higher education: The impact of college students' exposure to diversity on cultural awareness and political participation. *Journal of Education Policy*, 17(3):305-320.

DOI:10.1080/02680930210127577

Kamenetz, A. (2018). DeVos to rescind Obama-era guidance on school discipline. NPR. Retrieved from https://www.npr.org/2018/12/18/675556455/devos-to-rescindobama-era-guidance-on-school-discipline.

- Kang-Brown, J., Trone, J., Fratello, J., & Daftary-Kapur, T. (2013). A generation later: what we've learned about zero-tolerance in schools. Retrieved from: http://www.vera.org/sites/default/files/resources/downloads/zero-tolerance-inschools-policy-brief.pdf
- Kewal, A., Gilbertson, L., Fox, M., & Provasnik, S. (2007). Status and trends in the education of racial and ethnic minorities (NCES 2007-039). National Center for Education Statistics, Institute of Education Sciences, U.S. Department of Education. Washington, DC.
- Kim, C. Y, Losen, D. J., & Hewitt, D. T. (2010). The school-to-prison pipeline: Structuring legal reform. New York: New York University Press.
- Kim, C. Y. (2009/10). Procedures for public law remediation in school-to-prison pipeline litigation: Lessons learned from Antoine v. Winner School District. New York *Law School Review*, 54, 956-974.
- Kinsler, J. (2011). Understanding the black-white school discipline gap. *Economics of Education Review*, 30(6), 1370–1383.
- Koon, D. (2013, April). Exclusionary school discipline: An issue brief and review of literature. Retrieved from https://www.law.berkeley.edu/files/ BMOC\_Exclusionary\_School\_Discipline\_Final.pdf
- Kupchik A., & Ellis, N. (2008). School discipline and security: Fair for all students?*Youth Society*, 39(4), 549-574. doi:10.1177/0044118X07301956
- Ladson-Billings, G. (1994). *The dreamkeepers: Successful teaching for African-American students*. San Francisco: Jossey-Bass.

Ladson-Billings, G. (1998). Just what is critical race theory and what's it doing in a nice field like education? *International Journal of Qualitative Studies in Education*, *11*(1), 7-24.

DOI:10.1080/095183998236863

- Ladson-Billings, G. (1999). Just what is critical race theory, and what's it doing in a nice field like education? Race Is...Race Isn't: Critical Race Theory and Qualitative Studies in Education. (pp. 7-30). Boulder, Co.: Westview Press.
- Ladson-Billings, G. J. (1999). Chapter 7: Preparing teachers for diverse student populations: A critical race theory perspective. *Review of Research in Education*, 24(1), 211–247. https://doi.org/10.3102/0091732X024001211
- Ledesma, M. C., & Calderón, D. (2015). Critical race theory in education: A review of past literature and a look to the future. *Qualitative Inquiry*, 21(3), 206–222. https://doi.org/10.1177/1077800414557825
- Losen, D. J, & Martinez, T. E. (2013). Out of school and off track: The overuse of suspensions in American middle and high schools. UCLA: The Civil Rights Project / Proyecto Derechos Civiles. Retrieved from https://escholarship.org/uc/item/8pd0s08z
- Losen, D. J., & Skiba, R. J. (2010). Suspended education: Urban middle schools in crisis. Montgomery, AL: Southern Poverty Law Center. Retrieved September 19, 2012, from http://www.splcenter.org/ get-informed/publications/suspended-education
- Losen, D. J., Keith, M. A., II, Hodson, C. L., Martinez, T. E., Belway, S., & Civil Rights Project, P. D. C. C. for C. R. R. (CCRR). (2015). Closing the school discipline

gap in California: Signs of progress. In Civil Rights Project - Proyecto Derechos Civils. Civil Rights Project - Proyecto Derechos Civils.

- Losen, D., Hodson, C., & Keith, M. (2015). Are we closing the school discipline gap? The Center for Civil Rights Remedies.
- Losen, D.L., & Gillespie, J. (2012). Opportunities suspended: The disparate impact of disciplinary exclusion from schools. The Civil Rights Project at UCLA. Retrieved from: http://civilrightsproject.ucla.edu/resources/projects/center-forcivil-rights-remedies/school-to-prison-folder/federal-reports/upcoming-ccrrresearch/losen-gillespie-opportunity-suspended-ccrr-2012.pdf
- Louviere, J. J., Hensher, D. A., & Swait, J. D. (2000). *Stated choice methods: analysis and applications*. Cambridge, United Kingdom: Cambridge University Press.
- Loveless, T. (2017). Racial disparities in school suspensions. *The Brown Center Chalkboard*. Retrieved from https://www.brookings.edu/blog/brown-centerchalkboard/2017/03/24/racial-disparities-in-school-suspensions/
- Magnuson, K. A., & Duncan, G. J. (2006). The role of family socioeconomic resources in the black-white test score gap among young children. *Developmental Review*, 26(4), 365–399.
- Mansoor, S. (2017). The Q&A: Augustina Reyes. *The Texas Tribune*. Retrieved from https://www.texastribune.org/2017/02/15/q-augustina-reyes/
- Massar, M., McIntosh, K., & Eliason, B. (2015). Do out-of-school suspensions prevent future exclusionary discipline? Office of Special Education Programs. Retrieved from https://eric.ed.gov/?id=ED572039

- Maxime F. (2018). Zero-tolerance policies and the school to prison pipeline. Shared Justice. Retrieved from http://www.sharedjustice.org/domesticjustice/2017/12/21/zero-tolerance-policies-and-the-school-to-prison-pipeline
- McIntosh, K., Frank, J. L., & Spaulding, S. A. (2010). Establishing research-based trajectories of office discipline referrals for individual students. *School Psychology Review*, 39(3), 380–394.
- McLaren, P (2015). Pedagogy of Insurrection: From Resurrection to Revolution. New York, NY.
- McLaren, P. (2014). *Life in schools. An introduction to critical pedagogy in the foundations of education* (6th ed.). Herndon, VA: Paradigm Publishers.
- McLaren, P. (2015). *Life in schools: An introduction to critical pedagogy in the foundations of education (6th ed.).* Routledge.
- Mendez, L. M. R. (2003). Predictors of suspension and negative school outcomes: A longitudinal investigation. *New Directions for Youth Development, 2003*(99), 17–33. https://doi.org/10.1002/yd.52
- Mendez, L., & Knoff, H. (2003). Who gets suspended from school and why: A demographic analysis of schools and disciplinary infractions in a large school district. *Education and Treatment of Children*, 26(1), 1-23. Retrieved from http://eds.a.ebscohost.com com/eds/pdfviewer/pdfviewer?vid=1&sid=dd91d1a9-3e41-4163-9a36-417b6b7dc0d9%40sdc-v-sessmgr03
- Mendez, L. M., Knoff, H. M., & Ferron, J. M. (2002). School demographic variables and out-of-school suspension rates: A quantitative and qualitative analysis of a large,

ethnically diverse school district. *Psychology in the Schools, 39*(3), 259. https://doi.org/10.1002/pits.10020

Menard, S. (2009). *Logistic regression: From introductory to advanced concepts and applications*. Thousand Oaks, CA: SAGE

Miller, R. (2019). Betsy DeVos uses a racist research study to defend rescinding Obamaera discipline guidance. *The Progressive*. Retrieved from https://progressive.org/public-school-shakedown/betsy-devos-uses-racistresearch-study-miller-190411/

- Milner, H. R. (2013). Why are students of color (still) punished more severely and frequently than white students? *Urban Education*, 48(4), 483–489. https://doi.org/10.1177/0042085913493040
- Morgan, E., Salomon, N., Plotkin, M., A., & Cohen, R. (2014). The school discipline consensus report: Strategies from the field to keep students engaged in school and out of the juvenile justice system. Retrieved from http://csgjusticecenter.org/wp content/uploads/2014/06/The\_School\_Discipline\_Consensus\_Report.pdf
- Morgan, S., Plotkin, N., Plotkin, M., & Cohen, R. (2014). School discipline consensus report: Strategies from the field to keep students engaged in school and out of the juvenile justice system. Office of Juvenile Justice and Delinquency Prevention (OJJDP. Retrieved from

https://www.ncjrs.gov/App/Publications/abstract.aspx?ID=271355

Morris, E. W., & Perry, B. L. (2017). Girls behaving badly? Race, gender, and subjective evaluation in the discipline of African American girls. *Sociology of Education*, 90(2), 127–148. https://doi.org/10.1177/0038040717694876

- Mowen, T., & Brent, J. (2016). School discipline as a turning point: The cumulative effect of suspension on arrest. *Journal of Research in Crime and Delinquency*, 53(5), 628–653. https://doi.org/10.1177/0022427816643135
- Murnane, R. J. (2013). U.S. high school graduation rates: Patterns and explanations. *Journal of Economic Literature*, 51(2), 370–422. https://doi.org/10.1257/jel.51.2.370
- Musu-Gillette, L., Robinson, J., McFarland, J., KewalRamani, A., Zhang, A., &
  Wilkinson-Flicker, S. (2016). Status and trends in the education of racial and ethnic groups 2016. U.S. Department of Education, National Center for Education Statistics. Washington, DC. Retrieved from http://nces.ed.gov/pubsearch
- Musu-Gillette, L., Zhang, A., Wang, K., Zhang, J., Kemp, J., Diliberti, M., & Oudekerk,
  B.A. (2018). Indicators of school crime and safety: 2017. National Center for
  Education Statistics, U.S. Department of Education, and Bureau of Justice
  Statistics, Office of Justice Programs, U.S. Department of Justice. Washington,
  DC.
- Nasir, N. S. (2012). Racialized identities: Race and achievement among African American youth. Stanford, CA: Stanford University Press. Retrieved from http://www.questia.com

National Center for Education Statistics. (2016). *Percentage of public school students in* grades 6 through 12 who had ever been suspended or expelled, by sex and race/ethnicity: Selected years, 1993 through 2012 (School Safety and Discipline Survey and Parent and Family Involvement in Education Survey of the National Household Education Surveys Program No. Table 233.20). Washington, D. C.: United States Department of Education.

- Nicholson-Crotty, S., Birchmeier, Z., & Valentine, D. (2009). Exploring the impact of school discipline on racial disproportion in the juvenile justice system. *Social Science Quarterly*, 90(4), 1003–1018. https://doi.org/10.1111/j.1540
- Racial and Gender Disproportionality in School Punishment. The Urban Review, 34(4), Research Journal, 51(4), 640-670.
- Noltemeyer, A. L., Ward, R. M., & Mcloughlin, C. (2015). Relationship between school suspension and student outcomes: A meta-analysis. *School Psychology Review*, 44(2), 224-240.
- Noguera, P. A. (2003). Schools, prisons, and social implications of punishment: Rethinking disciplinary practices. *Theory Into Practice*, 42(4), 341–350. https://doi.org/10.1207/s15430421tip4204
- Noguera, P., Bishop, J. Howard, T & Johnson, S. (2019). Beyond the schoolhouse:
  Overcoming challenges & expanding opportunities for black youth in Los
  Angeles County. Center for the Transformation of Schools, Black Male Institute,
  Graduate School of Education & Information Studies, University of California,
  Los Angeles.
- Nowick, J. (2018). K-12 Education: Discipline disparities for black students, boys, and students with disabilities. Report To Congressional Requesters. Gao-18-258.
  Retrieved from https://eric.ed.gov/?id=ED590845
  NYU Press.

- Office for Civil Rights, "2015–16 Civil Rights Data Collection: School Climate and Safety" (Washington: U.S. Department of Education, 2018), available at https://www2.ed.gov/about/offices/list/ocr/docs/school-climate-and-safety.pdf
- Paige, R., & Witty, E. (2010). *The black-white achievement gap: Why closing it is the greatest civil rights issue of our time*. New York: AMACOM.
- Pavelka, S. (2013). Practices and policies for implementing restorative justice within schools. *Prevention Researcher*, 20(1), 15–17.
- Payne, A., & Welch, K. (2010). Modeling the effects of racial threat on punitive and restorative school discipline practices. *Criminology*, 48(4), 1019–1062. https://doi.org/10.1111/j.1745-9125.2010.00211.x
- Pearman, F. A., Curran, F. C., Fisher B., & Gardella, J. (2019). Are achievement gaps related to discipline gaps? Evidence from National Data. *AERA Open*. October 2019. doi:10.1177/2332858419875440
- Rabaka, R. (2006). The souls of black radical folk: W. E. B. Du Bois, critical social theory, and the state of Africana studies. *Journal of Black Studies*. *36*(5),732-763. doi:10.1177/0021934705285941
- Rafa, A. (2019, January). The status of school discipline in state policy. *Education Commission of the States*. Retrieved from https://www.ecs.org/wp-content/uploads/The-Status-of-School-Discipline-in-State-Policy.pdf
- Raffaele Mendez, L. M., & Knoff, H. M. (2003). Who gets suspended from school and why: A demographic analysis of schools and disciplinary infractions in a large school district. *Education and Treatment of Children*, 26(1), 30–51.

- Raffaele Mendez, L. M., Knoff, H. M., & Ferron, J. M. (2002). School demographic variables and out-of-school suspension rates: A quantitative and qualitative analysis of a large ethnically diverse school district. *Psychology in the Schools,* 39, 259–277.
- Rausch, M. K., & Skiba, R. (2004). Unplanned outcomes: Suspensions and expulsions in Indiana. *Education Policy Briefs*, 2(2), 1-8.

Report 2013–2014. Retrieved from ritter.tea.state.tx.us/adhocrpt/

- Reyes, A. H. (2006). *Discipline, achievement, & race: Is zero tolerance the answer?* Lanham, MD: Rowman & Littlefield Education.
- Riddle, T., & Sinclair, S. (2019). Racial disparities in school-based disciplinary actions are associated with county-level rates of racial bias. *Proceedings of the National Academy of Sciences*, 116 (17) 8255-8260. DOI: 10.1073/pnas.1808307116
- Rivers, S. R. (2020). Expression of the oppressed: Using critical pedagogy in arts education to disrupt systems of oppression. *Visions of Research in Music Education*, 35(9), 1–11.
- Rocque, M. (2010). Office discipline and student behavior: Does race matter? *American Journal of Education*, *116*(4), 557–581. doi:10.1086/653629
- Rowley, R. L., & Wright, D. W. (2011). No "White" child left behind: The academic achievement gap between black and white students. *Journal of Negro Education*, 80(2), 93–107.
- Skiba, R. J. (2013). Reaching a critical juncture for our kids: The need to reassess schooljustice practices. *Family Court Review*, 51(3), 380–387. https://doi.org/10.1111/fcre.12034

- Skiba, R. J., & Losen, D. J. (2016). From reaction to prevention: Turning the page on school discipline. *American Educator*, 39(4), 4-11.
- Skiba, R. J., & Noam, G. G. (2001). Zero tolerance: Can suspension and expulsion keep schools safe? San Francisco, CA: Jossey-Bass.
- Skiba, R. J., & Peterson, R. L. (2000). School discipline at a crossroads: From zero tolerance to early response. *Exceptional Children*, 66(3), 335–346.
- Skiba, R. J., Michael, R. S., Nardo, A. C., & Peterson, R. L. (2002). The color of discipline: Sources of racial and gender disproportionality in school punishment. *Urban Review*, 34, 317–342.
- Skiba, R. J., Trachok, M., Chung, C., Baker, T., & Hughes, R. (2012). Parsing disciplinary disproportionality: Contributions of behavior, student, and school characteristics to suspension and expulsion. Paper presented at the annual meeting of the American Education Research Association, Vancouver, BC.
- Skiba, R., & Williams, N. (2014, March). Are black kids worse? Myths and facts about racial differences in behavior. The Equity Project at Indiana University Center for Evaluation and Education Policy. Retrieved from http://www.indiana.edu/~atlantic/wp-content uploads/2014/03/African-American-Differential-Behavior\_031214.pdf
- Skiba, R. J., Horner, R. H., Chung, C.-G., Rausch, M. K., May, S. L., & Tobin, T. (2011).
  Race is not neutral: A national investigation of African American and Latino disproportionality in school discipline. *School Psychology Review*, 40(1), 85–107.

- Skiba, R., Michael, R., & Nardo, C. & Peterson (2002). The color of discipline: Sources of racial and gender disproportionality in school punishment. *The Urban Review*, 34 (4), 317–342. https://doi.org/10.1023/A:1021320817372
- Skiba. R.J, Chung, C., Trachok, M., Baker, T., Sheya, A., & Hughes, R. (2014). Parsing disciplinary disproportionality: Contributions of infraction, student, and school characteristics to out-of-school suspension and expulsion. *American Educational Research Journal*, 51(4), 640-670. https://doi.org/10.3102/0002831214541670
- Skiba. R.J, Chung, C., Trachok, M., Baker, T., Sheya, A., & Hughes, R. (2014). Parsing the nondiscriminatory administration of school discipline.
- Skiba, R. J., & Noam, G. G. (Eds.). (2001). Zero tolerance: Can suspension and expulsion keep schools safe? San Francisco, CA: Jossey-Bass.
- Skiba, R. J., Simmons, A. B., Ritter, S., Gibb, A. C., Rausch, M. K., & Cuadrado, J. (2008). Achieving equity in special education: History, status, and current challenges. *Exceptional Children*, 74, 264–288.
- Sleeter, C.E., & Delgado-Bernal, D. (2004). Critical pedagogy, critical race theory, and antiracist education. In J.A. Banks & C.A. McGee Banks (Eds), *Handbook of research on multicultural education* (pp.240-258). San Francisco: Jossey-Bass.
- Slowikowski, J. (2009). *Reducing disproportionate minority contact: Preparation at the local level.* Office of Juvenile Justice and Delinquency Prevention. https://www.ncjrs.gov/pdffiles1/ojjdp/218861.pdf
- Smith, C. D. (2009). Deconstructing the pipeline: Evaluating school-to-prison pipeline equal protection cases through a structural racism framework. *Fordham Urban Law Journal*, 36(5), 1009–1049.

- Solórzano, D. G., & Yosso, T. J. (2002). Critical race methodology: Counter-storytelling as an analytical framework for education research. *Qualitative Inquiry*, 8(1), 23–44. https://doi.org/10.1177/107780040200800103
- Stahl, S. D. (2016). The evolution of zero-tolerance policies. *CrissCross*, 4(1), Article 7. Retrieved from https://digitalcommons.iwu.edu/crisscross/vol4/iss1/7
- Steinberg, M., & Lacoe, J. (2017). What do we know about school discipline reform? *Education Next*, 17(1), 1-9. Retrieved from https://www.educationnext.org/whatdo-we-know-about-school-discipline-reform-suspensions-expulsions/
- Stevens, S., & Evans, N. (2018). Five reasons why American schools are still separate and unequal, 64 years after Brown v. Board. National Women's Law Center. Retrieved from https://nwlc.org/blog/five-reasons-why-american-schools-are-stillseparate-and-unequal-64years-after-brown-v-board/
- TABSE. (2019). The Journal of the Texas Alliance of Black School Educators.
- Tamborini, C. R., Kim, C., & Sakamoto, A. (2015). Education and lifetime earnings in the United States. *Demography*, 52(4), 1383–1407. https://doi.org/10.1007/s13524-015-0407-0
- Taylor, M. C., & Foster, G. A. (1986). Bad boys and school suspensions: Public policy implications for Black boys. *Sociological Inquiry*, 56, 498–506.
- Texas Appleseed. (2007). Texas' school to prison pipeline: Dropout to incarceration. The impact of school discipline and zero-tolerance. Austin, Texas: *Texas Appleseed*.
- Texas Appleseed. (2010). Texas' school to prison pipeline: Expulsion. The path from lockout to dropout. Austin, Texas: *Texas Appleseed*.

- Texas Appleseed and Texans Care for Children. (2016). Dangerous discipline: How Texas schools are relying on law enforcement, courts, and juvenile probation to discipline students. (Report). Retrieved from http://stories.texasappleseed.org/dangerous-discipline
- Texas Appleseed. (2017-2018). Texas: The state of school discipline dangerous discipline: How Texas schools (Report). Retrieved from https://www.texasappleseed.org/sites/default/files/SchoolDisciplineinTexasnew.pdf
- Texas Association of School Board. (2019). Exclusionary discipline in Texas schools. legal questions and concerns. Retrieved from https://www.tasb.org/services/legalservices/tasb-school-law-esource/students/documents/exclusionary\_discipline.pdf
- Texas Education Agency. (2018-2019). State level annual discipline summary PEIMS discipline data. Retrieved from

https://rptsvr1.tea.texas.gov/cgi/sas/broker?\_service=marykay&\_program=adhoc. download\_static\_summary.sas&district=&agg\_level=STATE&referrer=Downloa d\_State\_Summaries.html&test\_flag=&\_debug=0&school\_yr=19&report\_type=ht ml&Download\_State\_Summary=Next

- Texas Education Agency. (2014). Public education information management system (PEIMS)
- The Children's Defense Fund. (1975). School suspensions: Are they helping children? Cambridge, MA: Washington Research Project.

The School-to-Prison Pipeline. (2010, April 14). The Texas Tribune.

- The U.S. Department of Education Office for Civil Rights. (2014). Civil Rights data collection data snapshot: School discipline. Issue Brief No. 1 (March 2014). https://ocrdata.ed.gov/Downloads/CRDC-School-Discipline-Snapshot.pdf
- Townsend, B. L. (2000). The disproportionate discipline of African American learners:
  Reducing school suspensions and expulsions. *Exceptional Children*, 66(3), 381–391.
- U.S. Department of Education. (2015). Guiding principles: A resource guide for improving school climate and discipline. U.S. Department of Education. Retrieved from http://www2.ed.gov
- U.S. Department of Education, Office for Civil Rights, 2015-2016 Civil Rights Data Collection: School Climate and Safety (2018).

https://www2.ed.gov/about/offices/list/ocr/docs/school-climate-and-safety.pdf.

Ujifusa, A. (2018, December). Betsy DeVos revokes Obama discipline guidance designed to protect students of color. Education Week. Retrieved from http://blogs.edweek.org/edweek/campaign-k

12/2018/12/betsy\_devos\_revokes\_obama\_discipline\_guidance\_students\_of\_color \_protect.html

United States. President. (1993-2001: Clinton). (1995). Proposed legislation: "The Gun-Free School Zones Amendments Act of 1995": message from the President of the United States transmitting a draft of proposed legislation to amend the Gun-Free School Zones Act of 1990 to provide the necessary nexus with interstate commerce. Washington: U.S. G.P.O.

- Vara-Orta, F. (2018). DeVos has axed Obama discipline guidelines meant to reduce suspensions of students of color. Retrieved from https://chalkbeat.org/posts/us/2018/12/21/its-official-devos-scraps-obamadiscipline-rules meant-to-reduce-suspensions-of-students-of-color/
- Viadero, D. (2020). Policy and practice diverge in disparate ways when it comes to school expulsion. *Education Week*, 39(27). Retrieved April 1, 2020, from http://blogs.edweek.org./edweek/insideschoolresearch/2020/03/bygabriellewanne hwhile.html
- Waitoller, F. R., & King Thorius, K. A. (2016). Cross-pollinating culturally sustaining pedagogy and universal design for learning: toward an inclusive pedagogy that accounts for dis/ability. *Harvard Educational Review*, *86*(3), 366-389
   DOI:10.17763/1943-5045-86.3.366
- Wald, J., & Losen, D. J. (2003). Defining and redirecting a school-to-prison pipeline. New Directions for Youth Development, 2003(99), 9–15. https://doi.org/10.1002/yd.51
- Walker, B., & Townsend, L. (2014, Fall). Sixty years after "Brown v. Board of Education legal and policy fictions in school desegregation, the individuals with disabilities education act, and no child left behind. *Multiple Voices for Ethnically Diverse Exceptional Learners, 14*(2), 41-51. Retrieved from http://eds.a.ebscohost.com/eds/pdfviewer/pdfviewer?vid=1&sid=d4a40d3d-bd5b-
- Welsh, R. O., & Little, S. (2018). The school discipline dilemma: A comprehensive review of disparities and alternative approaches. *Review of Educational Research*, 88(5), 752–794. https://doi.org/10.3102/0034654318791582

- White, C. (2014). Ian Haney López on the dog whistle politics of race (Part I) [Video file]. Retrieved from https://billmoyers.com/episode/ian-haney-lopez-on-the-dog-whistle-politics-of-race/
- Wise, S. (2016). Virginia student charged with stealing 'free' milk from school. https://wtvr.com/2016/05/23/free-milk-theft/

Zammit, S., Routitsky, A., & Greenwood, L. (2002). Mathematics and science achievement of junior secondary school students in Australia. . https://research.acer.edu.au/timss\_monographs/8